ANALYSING THE RELATIONSHIP BETWEEN GOVERNMENT EXPENDITURE IN AGRICULTURE, THE VALUE OF AGRICULTURAL PRODUCTION, AND OTHER SELECTED VARIABLES IN SOUTH AFRICA FOR THE PERIOD 1983-2019

ΒY

ETIAN NGOBENI

A MINI-DISSERTATION Submitted in partial fulfilment of the requirements for the degree

of

Master of Science

in

Agricultural Economics

in the

FACULTY OF SCIENCE AND AGRICULTURE

(School of Agricultural and Environmental Sciences)

at the

UNIVERSITY OF LIMPOPO, SOUTH AFRICA

SUPERVISOR: DR L.C MUCHOPA

DECLARATION

I, Etian Ngobeni, declare that the mini-dissertation entitled: ANALYSING THE RELATIONSHIP BETWEEN GOVERNMENT EXPENDITURE IN AGRICULTURE, THE VALUE OF AGRICULTURAL PRODUCTION, AND OTHER SELECTED VARIABLES IN SOUTH AFRICA FOR THE PERIOD 1983-2019: submitted to the University of Limpopo in partial fulfilment for the requirement of degree on Master of Science in Agricultural Economics, has not been submitted before at the University of Limpopo or any other institution and all sources or materials used have been duly acknowledged.

05/10/2022

Signature

Date

DEDICATION

I sincerely dedicate this mini-dissertation to my parents Elinah Ngobeni and Jonas Ngobeni, My grandmother Elizabeth Ngobeni, and my aunt Mphephu Ngobeni.

ACKNOWLEDGMENTS

I appreciate God for being the source of my strength, courage, and wisdom in this study. I thank God for loving me so much, blessing me with an opportunity to embark on this research, and providing me with all the resources I used to complete this minidissertation.

To my supervisor Dr L.C Muchopa. Thank you for your excellent supervision, time, patience, and encouragement. Your work ethic and dedication made it possible for me to complete this mini-dissertation efficiently. I am grateful to have been under your guidance in conducting this study. I also thank Mr Aphane T.R for all the academic discussions that made this journey enjoyable.

I am grateful for the financial support from the National Research Foundation of South Africa. Completing this study would have been impossible without the provided financial assistance.

To my parents, Elinah Ngobeni and Jonas Ngobeni, thank you for your endless love and support. To my grandmother, Elizabeth Ngobeni, and aunt Mphephu Ngobeni, thank you for loving me and always being there for me. I thank my whole family and all my friends for the endless love which kept me motivated throughout the period of completing this study.

ABSTRACT

Agricultural production measures the performance and efficiency of a country's agricultural sector. The state of agricultural production can be assessed through the value of agricultural production, which is a product of agricultural gross production and output prices in monetary terms. The study examines the relationship between the value of agricultural production, government spending on agriculture, and other selected variables. Annual data for the value of agricultural production, government expenditure in agriculture, consumer price index, average annual rainfall, food import value, and population from 1983 to 2019 were collected from different sources and were used in the analysis for this study.

The Johansen cointegration test was used to determine the existence of a long-run relationship between the value of agricultural production and selected variables by using both the trace and eigenvalue tests. The results indicated that there is a long-run relationship among the variables. The study further used the Granger causality test to check the causality between the value of agricultural production and government expenditure in agriculture. The results show that there is no causal effect between the two variables. Lastly, the study used a Vector autoregressive (VAR) model to determine the relationship between the value of agricultural production and selected variables. The results of the VAR model indicated that government expenditure, average annual rainfall, food import value, and population positively affect the value of agricultural production. The study also found that the consumer price index negatively affects the value of agricultural production.

The study recommends that the government increase its spending on the agricultural sector, which could be in the form of research investment in technologies such as climate-smart agricultural technologies. Additionally, the study recommends that policymakers should review the monetary policy of South Africa to ensure price stability and prevent inflation. Lastly, the study recommends that the South African agricultural government should discourage imports and encourage South African agricultural producers to produce more major imported food products.

Keywords: Value of agricultural production, Government expenditure in agriculture, Johansen cointegration, Granger causality, VAR

iv

TABLE OF CONTENT

DECLARATION	i
DEDICATION	ii
ACKNOWLEDGMENTS	iii
ABSTRACT	iv
TABLE OF CONTENT	v
LIST OF TABLES	viii
LIST OF FIGURES	ix
LIST OF ACRONYMS	x
LIST OF APPENDIXES	xii
CHAPTER 1	1
INTRODUCTION	1
1.1 Background of the study	1
1.2 Problem statement	4
1.3 Rationale of the study	5
1.4 Scope of the study	6
1.5 The importance of the study	7
1.6 Structure of the study	8
CHAPTER 2	9
OVERVIEW AND HISTORIC TRENDS OF SELECTED VARIABLES IN SC AFRICA	OUTH 9
2.1 Introduction	9
2.2 Trends of selected variables	9
2.3 Summary	
CHAPTER 3	
LITERATURE REVIEW	17
3.1 Introduction	17
3.2 Definition of key concepts	17
3.3 Nature of agricultural production	
3.4. Government expenditure in agriculture in South Africa	19
3.5 The Keynesian theory	
3.6 Wagner's theory	21
3.7 The role of government expenditure on agricultural development	

3.8 Review of empirical studies on agricultural productivity and growth in South Africa	.23
3.9 Review of empirical studies on agricultural productivity and growth internationa	ally 23
3.10 Variables affecting the value of agricultural production	.24
3.11 Summary	.27
CHAPTER 4	28
RESEARCH METHODOLOGY	. 28
4.1 Introduction	.28
4.2 Study area	.28
4.3 Data type	.29
4.4 Data collection	29
4.5 Analytical techniques	30
4.5.1 Unit root testing	.31
4.5.2 The choice of lag length for VAR	.31
4.5.3 Cointegration testing	32
4.5.4 Causality testing	33
4.5.5 Vector autoregression technique	33
4.6 Diagnostic tests	34
4.6.1 Autocorrelation	35
4.6.2 Stability test	35
4.6.3 Heteroscedasticity	35
4.6.4 Normality test	36
4.7 Impulse response analysis	37
4.8 Variance decomposition	37
4.9 Summary	37
CHAPTER 5	39
RESULTS AND DISCUSSION	39
5.1 Introduction	39
5.2. Descriptive statistics	39
5.3. Unit root results	40
5.4 Lag order selection criteria	44
5.5 Johansen cointegration results	45
5.6 Granger causality results	47
5.7 Vector autoregressive model results	50

5.8 Diagnostic test results	58
5.9 Impulse response analysis	62
5.10 Variance decomposition	69
5.11 Summary	74
CHAPTER 6	76
SUMMARY, CONCLUSION, AND POLICY RECOMMENDATIONS	76
6.1 Introduction	76
6.2 Summary of findings	76
6.3 Conclusions	77
6.4 Recommendations	78
6.5 Limitations of the study	79
6.6 Suggestions for further studies	79
REFERENCES	80
APPENDIXES	

LIST OF TABLES

Table 4.1	: Summary of used variables
Table 5.1	: Descriptive statistics
Table 5.2	: ADF test results
Table 5.3	: Lag order selection criteria
Table 5.4	: Johansen Cointegration test results (all variables)
Table 5.5	: Johansen Cointegration tests results (VAP&GEA)
Table 5.6	: Granger Causality results
Table 5.7	: VAR model results
Table 5.8	: Outsourced VAR model results
Table 5.9	: Residual Portmanteau test for autocorrelation results
Table 5.10	: Heteroscedasticity results on individual components
Table 5.11	: Joint Heteroscedasticity test results
Table 5.12	: VAR Residual Normality test results
Table 5.13	: Variance decomposition on InVAP
Table 5.14	: Variance decomposition on InGEA
Table 5.15	: Variance decomposition on InCPI
Table 5.16	: Variance decomposition on InAAR
Table 5.17	: Variance decomposition on InFIV
Table 5.18	: Variance decomposition on InPG

LIST OF FIGURES

Figure 2.1: Value of agricultural production (2016-2020)	10
Figure 2.2: Annual Government expenditure in agriculture trend (1982-2014)	11
Figure 2.3: Annual average rainfall in South Africa (1904-2015)	
Figure 2.4: Annual Consumer price index in South Africa	13
Figure 2.5: Annual food import value in South Africa	14
Figure 2.6: Annual population in South Africa	15
Figure 4.1: South African map	29
Figure 5.1: Graphical illustrations of variables in levels (1983-2019)	41
Figure 5.2: Graphical illustrations of variables in first difference (1983-2019)	42
Figure 5.3: Inverse roots of the AR characteristic polynomial	
Figure 5.4: Impulse responses to the value of agricultural production	63
Figure 5.4: Impulse responses to government expenditure in agriculture	
Figure 5.4: Impulse responses to consumer price index	65
Figure 5.4: Impulse responses to average annual rainfall	66
Figure 5.4: Impulse responses to food import value	
Figure 5.4: Impulse responses to population	68

LIST OF ACRONYMS

AAR	Annual Average Rainfall
ABARES	Australian Bureau of Agricultural and Resource Economics and
	Sciences
ADF	Augmented Dickey-Fuller
AGRISETA	Agricultural Sector Education Training Authority
AIC	Akaike Information Criterion
AR	Autoregressive
ARC	Agricultural Research Council
ARDL	Autoregressive Distributed Lag
CASP	Comprehensive Agricultural Support Program
CE	Cointegrating Equations
CPI	Consumer Price Index
DAFF	Department of Agriculture, Forestry, and Fisheries
DALRRD	Department of Agriculture, Land Reform, and Rural Development
DAO	Decentralised Autonomous Organisation
Df	Degrees of Freedom
FAO	Food and Agricultural Organisation
FDI	Foreign Direct Investment
FIV	Food Import Value
FPE	Final Prediction Error
GDP	Gross Domestic Product
GEA	Government Expenditure in Agriculture
HQ	Hannan-Quinn Information Criterion
ICT	Information and Communications Technology
IMF	International Monetary Fund
ITA	International Trade Administration
LR	Likelihood Ratio
mm	Millimetre
NT	National Treasury
OECD	Organisation of Economic Cooperation and Development
OLS	Ordinary Least Squares

Population
Rand
South African Reserve Bank
South African Weather Services
Schwarz Criterion
Statistics South Africa
United Nations
United States
Value of Agricultural Production
Vector Autoregressive
Vector Error Correction Model
World Bank
World Trade Organisation

LIST OF APPENDIXES

Appendix 1: Data	98
Appendix 2: Descriptive statistics	100
Appendix 3: Johansen cointegration (all variables)	101
Appendix 4: Johansen cointegration (VAP and GEA)	
Appendix 5: Granger causality	103
Appendix 6: Log transformed data	104
Appendix 7: VAR results	105
Appendix 8: Stability test results	
Appendix 9: Autocorrelation results	107
Appendix 10: Heteroscedasticity test results	
Appendix 11: Normality test results	109

CHAPTER 1 INTRODUCTION

1.1 Background of the study

1.1.1 The importance of agricultural production

According to various studies (Greyling et al., 2015; Bernstein, 2013; Mutaudeen and Hussainatu, 2014), agricultural production in South Africa has been fluctuating over the years. In some years, agricultural production was stable, while in other years it was increasing either at an increasing or decreasing rate. Agricultural production measures the performance and efficiency of a country's agricultural sector (Rahman and Anik, 2020). The level of agricultural production also gives an insight depth into the importance and relevance of the agricultural sector when compared with other sectors (Liebenberg et al., 2010). Moreover, agricultural production is considered the most significant primary condition for economic production and is a key to improving social development (Liu et al., 2021). The role of agricultural production in sustainable economic development is well established across countries (Ahmed et al., 2019). When agricultural production output declines, the prices of products at the farm-gate rise (Altson et al., 2009) as the availability of products declines and scarcity arises. An increase in the prices of commodities at farm-gate causes an increase in the value of agricultural production (explained further in section 1.1.2) given a scenario where production output is maintained or improved and prices are determined by market forces in the situation where the government does not intervene directly to influence prices (Karfakis et al., 2011). Therefore, the importance of agricultural production output lies in its contribution to determining the prices of agricultural products and ultimately the efficiency of the whole country's agricultural sector. An increase in agricultural production output ensures that small and medium-scale farmers get better opportunities which ultimately increases the incomes of the poorest groups in the society and further enhances the contribution of the agricultural sector to national economic growth (Brownson et al., 2012).

1.1.2 The value of agricultural production in South Africa

In South Africa, just like in most developing countries globally, food insecurity is a central concern in which citizens always fail to afford and access nutritious food (FAO, 2015). The value of agricultural production fluctuates from time to time due to changes

in economic factors such as government expenditure (Jefferson et al., 2022). A fluctuating value of agricultural production entail instant and rapid high and low prices of agricultural products at farm gate, and these fluctuations imply different effects to the livelihood of consumers and producers. In scenarios where the value of agricultural production instantly become high, prices of food at retail stores relatively become high and expose consumers to financial vulnerability. Consumers and producers have conflicting objectives in which consumers aim at buying agricultural products at lower prices while consumers aim at selling products at higher prices for profit maximisation (Vasić et al., 2019). Subsequently, a high value of agricultural production is advantageous for producers and detrimental to consumers. The value of agricultural production is found by multiplying the total agricultural production output by the prices of commodities at farm-gate (FAO, 2021) meaning that the value of agricultural production is gross agricultural output expressed in monetary terms. In 2003 and 2004, the South African total value of agricultural production was estimated at R72 million (Kirsten et al., 2009). According to DAFF (2020), the value of agricultural production in South African context is derived from the prices of field crops, horticultural crops, and livestock products. Some studies (Lencucha et al., 2020; MacDonald et al., 2015; Taiba, 2014) argue that prices of agricultural commodities should be raised for farmers' incentives to expand production. However, increasing the prices of agricultural commodities also has significant consequences as it makes staple foods to be more expensive for poor sections of the population. Staple food prices were extremely high in the periods 1991/2; 1994/5; 2002/3; and then again in 2008/9 (Kirsten, 2015). Increasing farm prices redistributes income towards the agricultural sector, where the savings ratio is lower than in the rest of the economy (Cervantes-Godoy and Dewbre, 2010). Moreover, raising agricultural prices leads to lagged wage increases in the other sectors of the economy and at least a temporary decline in industrial profits and the rate of industrial growth (Zainab and Umar, 2015). Agricultural production output, the value of agricultural production and government expenditure are all intertwined as further elaborated in section 1.1.3.

1.1.3 Government expenditure in agriculture in South Africa

Government expenditure on agriculture is globally accepted as a crucial technique for increasing agricultural production (Wielechowski, 2019). To stabilize the value of agricultural production, the government allocates a budget to the agricultural sector

(Tsakok and Bruce, 2007). Government budgets should be balanced among different economic sectors (Mainali, 2021). However, there might be no budget balance in most developing countries. In South Africa, the agricultural sector is barely treated as other economic sectors in budgetary policies (Meijerink and Roza, 2007) and thus resulting in many agricultural market failures. The share of government expenditure on agriculture in the total South African government expenditure is lower than the share of agriculture in the economy (Ebenezer et al., 2019). Government expenditure in agriculture accounted for 1% of the total government expenditure in 2019/20 (STATS SA, 2021). The South African agricultural sector contributed a share of 2.4% to the economy in the period 2020/21 (STATS SA, 2021). The allocation of government expenditure on agriculture is of essential importance when contemplating the importance of food produced through agriculture and the share it has on the whole economy. The total budget allocated to the agricultural sector increased from R1.7 billion in 2001/2002 to R5.3 billion in 2014/2015 (National Treasury, 2016). In 2016 it was recorded that the budget that is allocated to the agricultural sector had stabilized. However, it was projected that the pressure on the fiscal policy was set to continue to limit any additional expenditure in the agricultural sector by the government for the years after 2016, which would likely affect service delivery. The government had a total budget of R1.9 trillion in 2019/20, and the agricultural sector received a share of R24 billion (National Treasury, 2021).

The South African government expenditure was projected to increase at an average annual rate of 4.5% from R6.8 billion in 2017 to R7.8 billion in 2020/2021 (STATS SA, 2021). The government funds about 80% of the country's agricultural research to find new agricultural strategies for a better level of production output. Moreover, 56% of the projected expenditure was for entities and conditional grants to provinces, which increased at an annual rate of 4.6% from R3.8 billion in 2017 to R4.4 billion in 2020/2021 (STATS SA, 2021). In the South African agricultural sector, compensation of employees takes R7 billion of the total expenditure over the medium-term. Additionally, the agricultural sector receives an additional R40 million to upgrade infrastructure and necessary equipment for analytical services laboratories in Pretoria and Stellenbosch (ARC, 2019).

1.1.4 Determinants of government expenditure in South Africa

Countries spend for different reasons, one of which is economic growth (Oladele et al., 2017). Government expenditure and its patterns are determined by complex factors as financial conditions and cultural, political, and economic factors (Fan and Rao, 2003). There are drivers of government expenditure in the economy and specific sectors. Some of the drivers of government expenditure include gross domestic product, government revenue, government debt, and fixed capital formation by the government (Mdluli et al., 2019). The ability of a country to spend depends on its willingness and ability to borrow from domestic and international sources and its own capacity to generate revenues (IMF, 2021). Maluleke (2018) identified population growth, poverty reduction, wage rate, trade openness, urbanisation, national income, and inflation rate as the drivers of government expenditure in South Africa. For government expenditure to be useful, it should be allocated to sectors such as agriculture since drivers of government expenditure emerge at the sectoral level (Zavale et al., 2011). While there are drivers of government expenditure, there are also obstacles that negate governments from spending (Wu et al., 2010). Mdluli et al. (2019) records that some of the obstacles against government expenditure in South Africa include the high unemployment rate which further reduces the revenue generated within the economy. Government expenditure is influenced by the need to eliminate economic inefficiencies that are caused by market failures (Curristine et al., 2007).

1.2 Problem statement

Economic growth in less developed countries is enhanced using government expenditure as an important instrument (Mapfumo *et al.*, 2012) and this translates as well to sectoral expenditure and growth. With the same spending goals for agriculture and given the general consensus that the South African agricultural sector is poised for job creation (Allen *et al.*, 2021), a lot more has to be understood in the context of government spending in the agricultural sector and agricultural production output. Ebenezer *et al.* (2019) state that government expenditure in agriculture is of vital importance for agricultural productivity. Generally, it is anticipated that improved agricultural productivity will have a positive impact on a nation's food security. According to Iganiga and Unemhilim (2011), various factors influence the growth in the value of agricultural production. These factors compass education, infrastructure, inflation, rainfall, and industrialization. In the context of South Africa, empirical studies

on the determinants of the value of agricultural production are very few. Although a study by Ebenezer et al. (2019) analysed the relationship between agricultural government spending and agricultural production output, the research did not consider other factors indicated in Iganiga and Unemhilin (2011) except for industrialisation. This study also differs from Ebenezer *et al.* (2019) in that, the present study analyses the value of agricultural production instead of the focus of Ebenezer et al. (2019) on agricultural production output. The value of agricultural production in monetary terms at farm-gate is obtained by multiplying gross production with output prices (FAO, 2021). According to OECD (2020), the value of agricultural production in South Africa has been fluctuating. A fluctuating value of agricultural production negatively affects the value adding/processing sectors of agriculture as well as the manufacturing sector and impedes economic growth. Therefore, it is important to monitor the movement of that variable (value of agricultural production) in time. Additionally, a fluctuating value of agricultural production forces households to rely on cheaper less nutritious food (Karfakis et al., 2011). Moreover, a fluctuating value of agricultural production negatively affect disadvantaged farmers because most of them do not have enough financial resources to sustain such unpredictability (Huka et al., 2014). This study will, therefore, fill the gap in knowledge regarding the fluctuations of the value of agricultural production over time by adding literature on the analysis of various factors that influence the value of agricultural production with emphasis on government spending in agriculture in South Africa.

1.3 Rationale of the study

Agriculture plays a significant role in the process of economic development and can significantly contribute to household food security in South Africa (DAFF, 2019). According to STATS SA (2020), a census conducted in 2017 on commercial agriculture showed that South Africa has 40122 highly capitalised farms. The smallholder sector comprises an estimated 1 762 000 smallholder farmers (Von Loeper and Blignaut, 2018). The agricultural sector is known for generating jobs, poverty eradication, and improving the livelihoods of rural households in the Sub-Saharan Africa region (FAO, 2016). Employment in the South African agricultural sector is estimated to be about 5% of the total labour force (World Bank, 2020). With this notion, the South African government has introduced various policies and programs intending to strengthen the agricultural sector to continue doing its

significant economic role. One of the policies is a fiscal policy which is a tool of economic management which encourages the use of public spending to achieve stability in the economy (Edeh *et al.*, 2020). The Fiscal policy involves governments' use of its public expenditure and revenue master plans to attain advantageous effects while negating undesirable effects on the county's employment, gross output, and productivity.

Government spending on agriculture is aimed at enhancing agricultural production. Omabitan and Khanal (2022) identified government spending as one of the main drivers of farm production output. Development economists consider production growth in the agricultural sector as crucial if the agricultural output is to increase significantly at an adequately expeditious rate to meet the growing demand for food (Black and Gerwel, 2014). The inability to analyse the impact of government spending can result in the negative effect of ineffective spending overwhelming the positive effects (Devarajan *et al.*, 1996). In South Africa, several studies (Chipaumire *et al.*, 2014; Molefe and Choga, 2017; Odhiambo, 2015), have analysed the impact of government spending on the overall economy neglecting to examine the effect it has on public sectors like agriculture. According to Moreno-Dodson (2008), government spending may be unproductive and further reduce the production ability of subsectors. Hence it is necessary to analyse the effect of government spending in agriculture on the value of agricultural production in South Africa, given that there is limited literature for the period evaluated in this study.

1.4 Scope of the study

1.4.1 Aim of the study

The study aims to examine the relationship between the value of agricultural production, government spending in agriculture, annual average rainfall, consumer price index, food import value, and population.

1.4.2 Objectives of the study

The objectives of this study are to:

I. Analyse the short-run and long-run relationship between government spending in agriculture and the value of agricultural production.

- II. Determine the causality between government spending in agriculture and the value of agricultural production.
- III. Determine the relationship between the value of agricultural production and selected independent variables: government spending in agriculture, annual average rainfall, consumer price index, food import value, and population.

1.4.3 Hypotheses

- I. There is no significant long-run and short-run relationship between government spending in agriculture and the value of agricultural production.
- II. There is no causality between government spending in agriculture and the value of agricultural production.
- III. There is no relationship between the value of agricultural production and selected independent variables: government spending in agriculture, annual average rainfall, consumer price index, food import value, and population.

1.5 The importance of the study

The South African agricultural sector contributes significantly to the livelihood of people and economic growth. As South Africa continues to work towards economic development, new knowledge on factors that influence economic growth at the sectoral level is needed. This study explores factors such as government expenditure in agriculture, consumer price index, annual average rainfall, food import value, and population which affect the value of agricultural production in South Africa. Therefore, this study will fill the gap in the literature and add to the studies that have not considered the effect of the aforementioned factors towards the value of agricultural production. Additionally, this study will give necessary information to the government policy makers on the impact that government spending in agriculture had on the value of agricultural production in South Africa from 1983 to 2019. Furthermore, the study will also give detailed information on the impact of other factors such as consumer price index, average rainfall, food import value, and population growth towards the value of agricultural production in South Africa. The government will get to understand how it should spend towards the agricultural sector in the future to enhance the value of agricultural production.

1.6 Structure of the study

This study is made up of six chapters in with which chapter 1 provides the introduction. Chapter 2 gives a historic trend analysis of the selected variables in South Africa. Chapter 3 gives detail into the theoretical and empirical literature related to the study. Chapter 4 explores the methodologies that are used in this study. In chapter 5, results and empirical analysis of the findings are discussed. Lastly, chapter 6 gives a summary and concludes the findings. Furthermore, chapter 6 gives an overview of the recommendations together with the limitations of the study and future research suggestions.

CHAPTER 2

OVERVIEW AND HISTORIC TRENDS OF SELECTED VARIABLES IN SOUTH AFRICA

2.1 Introduction

In this chapter, an overview of the selected variables relevant to this study is given based on available literature. Historic trends in South Africa on the selected variables that affect the value of agricultural production are illustrated using graphs and diagrams. Furthermore, explanations of the graphs and diagrams are given in this chapter. Since these variables affect the value of agricultural production, it is crucial to analyse their historic trends/behaviour.

2.2 Trends of selected variables

2.2.1 Value of agricultural production

The value of agricultural production in South Africa was R15 000 million in 1987 (Kirsten *et al.*, 1994). Even though the South African value of agricultural production was fluctuating in historical years, it increased between the period 1999 to 2018. DAFF (2017) recorded that the value of agricultural production in 2015 was R233 237 million compared to R220 983 million in 2014 which was an increase of 5.5%. The value of agricultural production increased by 15.9% in 2020 from a value of R287 395 million to a value of R332 953 million (DALRRD, 2020). The increase in the value of agricultural production between 2019 and 2020 is associated with an increase in the value of field crops.



Figure 2.1: Value of agricultural production (2016-2020)

Source: DALRRD (2020)

Figure 2.1 shows the value of agricultural production based on field crops, horticulture, and animal products for the period 2016 to 2020. According to figure 2.1, animal products contribute a larger share towards the value of agricultural production as compared to field crops and horticulture in South Africa. From 2016 to 2020, animal products have contributed more than R120 000 million towards the gross value of agricultural production in each year. Subsequently, Horticulture has contributed between R80 000 million and R120 000 million value of production towards the gross value of agricultural production in the period 2016 to 2020. Moreover, field crops have contributed less than R80 000 million towards the gross value of agricultural production in South Africa between 2016 and 2020. Meaning that field crops contribute less towards the value of agricultural products. According to DALRRD (2020), the gross value of animal products contributed 43,8% to the total gross value of agricultural production, while horticultural products and field crops contributed 30,5% and 25,7%, respectively.

2.2.2 Government expenditure in agriculture

In South Africa, the total government expenditure on agriculture inclusive of forestry and fisheries has expanded almost threefold between 2003 and 2014 (Visser and Ferrer, 2015). There was an annual increase of 10.8% which is from R5.77 billion in 2003 to R16.97 billion in 2014 (World Bank, 2015). From 2003, government expenditure on agriculture has increased by 15% annually until 2008/9. After the 2008/9 period, the growth resumed by 8.9% annually until 2014. Figure 2.2 shows the growth trend in government expenditure in agriculture from 1982 to 2014.



Figure 2.2: Annual Government expenditure in agriculture trend (1982-2014) Source: SARB, 2021

Government expenditure in agriculture has been fluctuating between 1982 and 2014 in South Africa based on figure 2.2. From 2004/5, the government introduced the Comprehensive Agricultural Support Program (CASP) to promote the provision of post-settlement support services to land reform beneficiaries which enhanced agricultural businesses (DAO, 2004). Since CASP was implemented, an aggregated amount of R750 million has been allocated to this program (DALRRD, 2021). This led to a stronger focus on agricultural rural development and farmer settlement and

financing. Hence the trend of government expenditure in agriculture after 2004 started to increase gradually even though it would decline in some years and instantly increase.

2.2.3 Average annual rainfall

Rainfall in South Africa is variable spatially and temporally (Morris *et al.*, 2005). Most of the rain in South Africa is received in the summer seasons. The average annual rainfall from 1936 to 1999 was 450 mm (Tennant and Hewitson, 2002). Rainfall was above average in the 1970s, the late 1980s, and mid to late 1990s, and below average in the 1960s and the early 2000s, reverting to average towards 2010 (Kruger and Nxumalo, 2017).



Figure 2.3: Annual average rainfall in South Africa (1904-2015) Source: SAWS (2016)

Figure 2.3 shows the average annual rainfall of South Africa for the period 1904 to 2015 and the bars show the average rainfall per year. Based on figure 2.3 it is evident that the rainfall in South Africa is variable as in some years it goes above the average level while in other years it goes below the average level. However, the South African

rainfall based on figure 2.3 shows below-average rainfall in many of the years covered in that figure.

2.2.4 Consumer price index

In the early 1980s, the consumer price index was low in South Africa (Moritz, 1994). The agricultural sector faced debt and high inflation in the early 1980s which harmed the whole economy (Vink and Van Rooyen, 2009). South Africa faced a severe drought in the early 1990s which influenced the consumer price index to be relatively high. Food prices in 1990 made an 18.6% weighting structure of the consumer price index from the early 1980s to 2020. The consumer price index is usually high due to factors such as drought which always put pressure on agricultural production output (Quiggin, 2008).



Figure 2.4: Annual Consumer price index in South Africa Source: DALRRD, 2021

Figure 2.4 shows that the South African consumer price index was increasing at a slow rate from the early 1980s to 1990. From 1990 the consumer price index started

to increase drastically until 2004 when it started stabilizing. However, in 2006 it started to increase, and then from 2008, it was increasing at an increasing rate due to the global downturn that was experienced in that period (Rena and Msoni, 2014). From 2011 to 2020, the consumer price index increased each year without declining or stabilizing.

2.2.5 Food import value

There are various reasons why South Africa imports food from other countries in different years (Potelwa *et al.*, 2016). The need to import might be a result of producing less agricultural output in a certain year (Lencucha *et al.*, 2020). If South Africa produces less food in a particular year, the chances are likely to be high for it to import more food, hence making the food import value to be relatively high and vice versa. As countries continue to develop, some are faced with less production due to resource scarcity (Dong *et al.*, 2022). Subsequently, countries are mandated to import from other countries, thus increasing the food import value. According to Kirsten *et al.* (1994), food imports in South Africa had a value of R1 200 million in 1993.



Figure 2.5: Annual food import value in South Africa Source: DALRRD, 2021

Figure 2.5 shows that the food import value in South Africa has been fluctuating from 1980 to 2020 because there is no consistent trend. Between some years the food import value was increasing while between some years it was decreasing. Food import value started increasing at an increasing rate from 2010 to 2017. Where it eventually declined in 2018 and picked up again from 2019 to 2020. The trend of food import value based on Figure 2.5 shows that South Africa grew in importing more agricultural commodities as a result of succumbing to food import value drivers such as low agricultural production output.

2.2.6 Population

The South African population was estimated to be 59.62 million by mid-2020, 62.3% of the population was aged between 16 and 59 (STATS SA, 2020). According to AGRISETA (2016), the agricultural sector employs most people in rural areas in South Africa. The sector employs an approximately 898 000 people, which represents 5.7% of the total labour force in the country.





Source: United Nations, 2019

Figure 2.6 indicates the actual population of South Africa from 1950 to 2019 and the projected annual population of South Africa from 2020 to 2100. Based on figure 2.6, in 1950 the population of South Africa was less than 20 million. Additionally, Figure 2.6 indicates that the population has been increasing without declining over the years. However, even though the trend increases from 1950 projected to 2075, the trend is predicted that it will stop increasing and continue to be stable from 2075 to 2100 as shown in Figure 2.6.

2.3 Summary

This chapter gave an overview and historic trends of selected variables in South Africa. In addition, the chapter made use of graphical illustrations for each selected variable and further outlined details of each graph in writing based on estimated values for the past years.

Since a review of selected variables was made in this chapter, the next chapter gives a detailed literature review of the study.

CHAPTER 3 LITERATURE REVIEW

3.1 Introduction

This chapter is divided into various sections in which the first one gives definitions to the key concepts used in this study. Additionally, the chapter explores theoretical perspectives related to this study. A theoretical discussion on general issues that are linked with agricultural production is outlined in this chapter. Furthermore, the chapter gives more details on theories of government expenditure and the role of government expenditure in agricultural development in both South African and international contexts. Variables that are assumed to be affecting the value of agricultural production are also explained in this chapter. An insight into national and international empirical studies related to this study is given. Finally, a summary of the whole chapter is outlined at the end.

3.2 Definition of key concepts

3.2.1 The value of agricultural production

According to FAO (2011), the value of agricultural production is the total output of agriculture expressed in monetary terms or average purchasing power parity agricultural prices. The value of agricultural production estimates the quality of agricultural production output at the prices with relation to the specified period (Carter, 1984). Hence, the value of agricultural production signifies the value of agricultural products at the time they were produced. This study adopts the definition of the value of agricultural production as stipulated by FAO (2011).

3.2.2 Government expenditure in agriculture

Government expenditure in agriculture depicts the expenses that are incurred by the government towards the agricultural sector (Atayi *et al.*, 2020). According to Mogues and Anson (2018), government expenditure in agriculture covers expenses such as agricultural research, technology development, livestock, crop gene banks, and extension services. Government expenditure in agriculture ensures that the agricultural sector becomes efficient in production which further helps the government

to achieve its economic objectives such as expanding employment opportunities (Awubare and Eyitope, 2015).

3.3 Nature of agricultural production

According to Liebenberg et al. (2011), the change in the size and structure of South Africa's agricultural sector and agricultural production dates back to the beginning of the 20th century. From the year 2000, the growth in crop production in South Africa has fallen as compared to the growth in the production of livestock which increased by 4.31% annually (Liebenberg et al., 2011). Agricultural production has become more important due to factors such as climate, population and food security (Ullah and Shivakoti, 2018). According to Oberč and Schnell (2020), most researchers in sustainable agriculture are trying to understand how to increase agricultural production by using measures that are both environmentally and economically friendly. Praburaj (2018) stipulates that expanding agricultural production and productivity leads to sustainable overall economic development, and it would be rational for governments to prioritise the development of the agricultural sector. Nevertheless, expanding agricultural production requires certain measures and policies such as government expenditure in agriculture. According to Matchaya (2020), understanding the essence of the causal process that occurs between the value of agricultural production output and government expenditure on agriculture helps to determine appropriate policy responses that can guarantee economic growth.

A study in Nigeria by Iganiga and Unemhilin (2011) analysed the impact of federal government agricultural spending on the value of agricultural production. The study used the Cobb-Douglas growth model, descriptive statistics and an econometric model to generate all empirical results. Consequently, the study found that public spending on agriculture has a positive impact on the value of agricultural production. In China, a study by Zeraibi and Mivumbi (2019) analysed the impact of public expenditure on the agricultural sector productivity using the gross output value of agriculture as a proxy variable for agricultural productivity. Additionally, the study used time series data from 1988 to 2018. To generate empirical results, the study used the Autoregressive Distributed Model (ARDL) which analysed the relationship between the value of agricultural sector. The study found that government expenditure in agriculture has a positive impact on the

gross output value of agriculture. Mohammed *et al.* (2017) conducted a study titled Gross Agricultural Production Valuable Land Labour Force in the Agricultural Sector and Causality of Energy Consumption in 76 countries using a dynamic panel data approach. The study used the granger causality test for empirical purposes and established that arable land and population granger cause agricultural gross production value. Ayoub and Michel (2019) analysed the impact of government spending on agricultural sector productivity in China using the value of gross agricultural output as a proxy variable for agricultural productivity. Subsequently, Ayoub and Michel (2019) used the Autoregressive Distributed Lag Model and the study found that there is a long-term relationship between government spending on agriculture and the value of gross agricultural output.

Most studies that are reviewed in this chapter are on agricultural productivity and agricultural production output since literature regarding the value of agricultural production is very limited in South Africa and internationally. Agricultural productivity measures the quantity of agricultural output that is produced given several inputs (ABARES, 2021). In addition, agricultural productivity is measured as the ratio of agricultural output to agricultural input (DAFF, 2011). Agricultural productivity expresses the total agricultural output in quantity form, while the value of agricultural production expresses the total agricultural output in monetary terms.

3.4. Government expenditure in agriculture in South Africa

Government expenditure is essential in the process of accumulating capital and improving long-term economic development (Gao *et al.*, 2022). According to Chen *et al.* (2019), the nature of government expenditure varies widely across countries and has undergone significant changes over time globally. Since 1994, the government of South Africa has implemented various initiatives aimed at supporting farmers (Khapayi and Celliers, 2016). Some of these initiatives include land reform and land redistribution for agricultural development which were implemented in 2000 (Hull *et al.*, 2019). In addition, the government implemented the Comprehensive Agricultural Support Program which focuses on the provision of post-settlement for black farmers, and the Micro-Agricultural Financial Institution of South Africa which provides micro-financing to emerging farmers and smallholder farmers who are not served by the Land Bank (DAFF, 2018). Even though the Micro-Agricultural Financial institution of

South Africa is under the department of agriculture, it operates under the Land Bank. The government allocates a budget to the agricultural sector every year, then the agricultural sector ultimately distributes the budget to different initiatives and organisations which are implemented. Finally, the initiatives and organisations are accessed by farmers at the local level to acquire financial resources. According to FAO (2009), a rise in the budget that the government allocates to the agricultural sector by 10% each year ensures that over 1.6 million farmers escape poverty by living above the poverty line. An increase in government expenditure in agriculture reduces production costs which leads to an increase in production and ultimately an increase in profitability of farms, and thus ensures economic growth (Nworji *et al.,* 2012).

3.5 The Keynesian theory

According to Biza *et al.* (2015), the Keynesian model stipulates that an increase in government expenditure plays a catalytic role in domestic economic activity and vibrant private investment. The Keynesian model suggests that government expenditure is an exogenous political tool that can be used to control economic activity in the short-run (Selvanathan *et al.*, 2021). Based on Keynes (1936), fiscal policy through public spending is seen as the main driver of economic growth. Hence government expenditure can be used as a strong tool that resolves all economic stagnation-related issues. However, the availability of too much government expenditure can be detrimental to private investments. Thus, South Africa is dependent mostly on foreign direct investments to sustain economic growth (Meniago and Peterson, 2013). The fact that South Africa depends on foreign direct investments means that government expenditure is not enough for maintaining all the public sectors including the agricultural sector.

According to Makin (2015), increasing government expenditure in a closed economy is followed by higher national output which further leads to more employment. Based on Keynes's theory, if governments stimulate government expenditure through fiscal policy, increased business activity will follow. And that is what most economies are always aiming to achieve. Keynes' theory further asserts that government spending improves aggregate output and paves the way for further increases in national income that result in economic growth. Public spending has a positive impact on economic growth, the causal relationship therefore runs from public expenditure to economic

growth (Keynes, 1936). However, Keynes (1936) was rejected by Wagner's law which considers public expenditure as an endogenous variable that can be used to stimulate the economy rather than as a cause of growth.

3.6 Wagner's theory

Wagner's law was the first government expenditure theory in the history of public finance (Maribe, 2020). The law stipulated that during the process of economic development, the ratio of public expenditure in the economy tends to increase at a higher rate than that of economic growth. Wagner (1876) gives three arguments for economic development as national revenue increases. Firstly, there are needs such as capital expenditure, redistribution of income, protection, and administration, which coexist with the process of development, and they are offered by the government. Secondly, an increase in welfare and cultural expenditures on public sectors such as education and agriculture take place when national income increases. Lastly, changes in technological structures and expansion of investments in economic activities for economic development would increase the number of private monopolies. Wagner's theory was initially well known among German economists before it was translated to English in 1958 (Manuel, 2018). According to Nyasha and Odhiambo (2019), Wagner's theory is the most cited law among other models of government expenditure growth in the world. Adolf Wagner made a hypothetical process in examining the trend in government expenditure growth where the hypothesis is that government spending does not play a crucial role in economic development.

On the contrary, economic growth improves government expenditure. Wagner's theory strongly believes that the causality between economic growth and government expenditure runs from economic growth to government expenditure and not vice versa as stated by the Keynesian Theory. Government expenditure will continue to stimulate economic growth to increase the well-being of its citizens. Furthermore, sustained government expenditure will result in increased investment in education, employment opportunities, a better quality of physical infrastructure, as well as sustained economic growth (Wagner, 1876). Wagner's theory is applicable in an economy where per capita income is the result of economic development. Sharma and Singh (2020) attempted to prove Wagner's theory in India in a study titled Evidence of Wagner's Law in India using time series data from 1960 to 2018. The study used a Vector Error Correction

Model in which government expenditure was treated as a dependent variable while gross domestic product per capita and urbanisation were treated as independent variables. The results of the Vector Error Correction Model showed that both urbanisation and the gross domestic product per capita have a positive and statistically significant impact on government expenditure in the long-run. Thus, proving that Wagner's theory holds in the economy of India. Moreover, Ibok and Bassey (2014) revisited Wagner's theory within the agricultural sector in Nigeria. Ibok and Bassey (2014) used the Johansen and Juselius cointegration test which revealed that there is a long-run relationship between agricultural capital expenditure and agricultural gross domestic product, the granger causality test proved Wagner's theory exists in the agricultural sector of Nigeria.

3.7 The role of government expenditure on agricultural development

As stated in the discussion of Keynesian theory, government expenditure is one of the most effective and prosperous measures to support economic growth. The economy is divided into different sectors and agriculture is one of them. Showing that the overall economy grows as sectors develop. The process of agricultural development generates a suitable space for realising the potential of the agricultural sector to the economy (Pawlak and Kołodziejczak, 2020). This is supported by the ideology that agricultural development is viewed as a continuous process involving the adoption of new agricultural production practices and inputs by farmers (Uremadu et al., 2018). Most developing countries tend to lose their participation ability in globalization due to less-developed agricultural sectors and low agricultural production which are uncompetitive in the global food market. In India, Selvaraj (1993) evaluated how the variation in public expenditure in agriculture affects the performance of the agricultural sector. The results showed that agricultural development depends on the long-term government budget allocated to the agricultural sector. Ogiogio (1995) conducted a study in Nigeria on the impact of government expenditure on total economic development for the period 1970 to 1993. The results showed that there is a significant relationship between government expenditure and economic growth. The study might be on economic development, but economic development is influenced by the growth in sectors such as agriculture. Showing that agricultural development depends critically on government expenditure. In both developing and developed countries, the

transformation of the agricultural sector depends on the investments made by the central governing body.

3.8 Review of empirical studies on agricultural productivity and growth in South Africa

In South Africa, Greyling (2015) conducted a study on analysing agricultural productivity and economic growth for the period 1970-2010 using the Engle-Granger two-step test. The results showed that the consumer price index negatively affects agricultural productivity while gross capital formation, gross domestic product, labour force, and agricultural exports positively affect agricultural productivity in the long run. A study by Kargbo (2007) on the analysis of the effect of macroeconomic factors towards agricultural productivity in South Africa for the period 1957 to 2004 used a Vector Error Correction Model. The study found that money supply, interest rate, exchange rate, and consumer price index have a positive significant effect on agricultural productivity. Furthermore, the study concluded that agricultural prices are the source of macroeconomic instability in South Africa. Setshedi (2019) conducted a study on an empirical analysis of macroeconomic variables related to agricultural productivity in South Africa for the period 1975 to 2016 using a Vector Error Correction Model. The results of the study showed that public spending and agricultural exports have a positive impact on agricultural productivity. Moreover, the consumer price index and the real effective exchange rate have a negative effect on agricultural productivity.

Reviewing South African empirical literature on the value of agricultural production is a challenge. Most studies that were conducted are on agricultural productivity and agricultural production output. Hence there are insufficient empirical studies for this topic. Therefore, this study analyses factors that affect the value of agricultural production with an emphasis on government expenditure on agriculture to add knowledge to the gap.

3.9 Review of empirical studies on agricultural productivity and growth internationally

In Nigeria, a study was conducted by Muhammad-Lawal and Atte (2009) analysing agricultural production using regression analysis and descriptive statistics, using the value of agricultural production as a proxy for agricultural production. The results
showed that food import value is detrimental to the value of agricultural production. Furthermore, the results showed that when government expenditure on agriculture increases, the value of agricultural production increases. Giang et al. (2019) conducted a study titled total factor productivity of agricultural firms in Vietnam and its relevant determinants. The study was conducted at the microeconomic level and productivity was measured as the total factor productivity of 420 agricultural enterprises. The study found that the determinants of agricultural productivity in Vietnam include size and age, the share of state and foreign ownership, export, accessibility to the internet and bank loan of firms. A study by Badar et al. (2007) in Pakistan on the factors affecting agricultural production used a Cobb-Douglas production function to generate empirical results. The results showed that expenditure on research is not significant while price support is significant towards agricultural production. Baba et al. (2010) assessed the impact of investment on agricultural growth and rural development in Himachal Pradesh of India. The study used the agricultural GDP as a proxy variable for agricultural productivity. Additionally, the study found that public and private investment in agriculture have a significant and positive relationship with agricultural productivity. Maïga et al (2021) analysed the effects of public expenditure on agricultural growth in Mali. The study used the Autoregressive distributed lag model and concluded that agricultural expenditure, sector employability and fertilizer consumption have significant negative effects and that education and health expenditure positively influence agricultural output. Jambo (2017) analysed the impact of government spending on agricultural growth in Zambia, South Africa, and Tanzania for the period 2000 to 2014. The study used the Vector Error Correction Model to test the impact of government spending, private investment and net trade on agricultural output growth. The results showed a negative relationship between agricultural growth and expenditure in Zambia. In Malawi, the results of the empirical analysis showed that agricultural research expenditure has a greater impact on agricultural growth, and, unlike in Zambia, there is evidence of a good relationship between agricultural growth and public expenditure. In addition, the results for South Africa showed that an increase in agricultural research expenditure increases agricultural growth.

3.10 Variables affecting the value of agricultural production

3.10.1 Government expenditure in agriculture

Expanding government expenditure in leading sectors such as agriculture drives economic activity and makes room for job creation (Ernawati *et al.*, 2021). Government expenditure in agriculture is responsible for the building of roads in rural areas, silos, tractors, and other equipment for farmers, which results in increased output and improved livelihoods of the people. Government expenditure on agriculture ensures the provision of loan facilities, subsidising of farm input, and financial backing to farmers which can make the agricultural sector additionally enticing and lift entrepreneurship in agribusiness, thereby resulting in positive externalities to different sectors of the economy. Regardless of any economic system in place, the role of government expenditure cannot be overemphasised. Government expenditure by any country is usually categorised into expenditure on security/defence, health, education, foreign affairs, administration, etc. (Aigheyisi, 2011). However, this study only focuses on government expenditure on agriculture.

3.10.2 Annual average rainfall

Water is regarded as one of the key elements for agricultural production, and therefore changes in water availability may have effects on agricultural production and the value of agricultural production (Torres et al., 2019). Rainfall is regarded as the best source of water globally. Moreover, rainfall is associated with economic growth and is crucial in dry and poor areas that depend on rainfed agriculture (Sangkhaphan and Shu, 2020). Hence the amount of rainfall is one of the most significant factors that affect agricultural production for farmers. Agricultural production is sensitive to changes in average annual rainfall (Tabosa, 2021). According to World Bank (2021), South Africa has an estimated annual rainfall of 450mm, which is below the average of 860mm. Rainfall poses risk and uncertainty in agricultural production for any season, especially in countries that practice rainfed agriculture, which is mainly developing countries like South Africa. A study by Amare et al. (2021) in Nigeria found that a negative rainfall shock strongly affects agricultural production negatively in which a negative rainfall shock reduces agricultural production by 38%. Furthermore, South Africa has warmer days than cooler days. Hence, rainfed agriculture is not reliable for most farmers and it is the reason why agricultural production is higher in irrigated areas than in rainfed places (Jaramillo et al., 2020).

3.10.3 Consumer price index

According to Keynes' theory, the consumer price index is a short-run outcome of demand pressure within the economy (Shilongo, 2019). The consumer price index is one of the most well executed and researched statistical programs globally. The consumer price index is openly the most followed measure of inflation. According to Ribba (2022), the consumer price index as a measure of inflation controls how governments spend. When the consumer price index increases, the value of agricultural production also increases (Muhammad-Lawal and Atte, 2009). Olatunji *et al.* (2012) used the Granger causality method and descriptive statistics in analysing agricultural production inflation in Nigeria. According to the study, changes in agricultural production (changes in stocks) resulted from changes in inflation for the years 1970 to 2006. This implies that the consumer price index and agricultural production are related or dependent on one another.

3.10.4 Food import value

Food imports are gradually becoming significant due to their positive effect on economic growth (Lee and Liu, 2014). According to Khashei and Bijari (2012), forecasting the value of agricultural food commodities must be done accurately because the changes in the agricultural sector and policy reforms affect the value of agricultural imports. Moreover, since agricultural development is a global mandate, governments must ensure that unnecessary costs are not incurred with the increasing food imports to minimize import security risks (Adewuyi *et al.*, 2021). Import security risks are risks associated with importing enough quantity of food at affordable prices and at the right time (Baranauskaite and Jureviciene, 2021). Agricultural production output has been fluctuating over the years and food imports in South Africa have been increasing exponentially (Greyling *et al.*, 2015). South Africa was the net exporter of agricultural products in the 1980s (Oluwatoba *et al.*, 2015). But the effects of sanctions on imports from South Africa due to the apartheid regime, as well as the inability to maintain international competitiveness, have prevented the country from maintaining its status as a net exporter of agricultural products (Linderson, 2020).

3.10.5 Population

Dao (2012) analysed the effect of population on economic growth using a statistical model. The study was based on 43 developing countries for the period 1990 to 2008. The results showed that an increase in population slows economic growth. In addition,

economists usually identify population growth as a driver of economic growth because an increase in the population entails an increase in the labour force (Olulu *et al.*, 2014). The history of Northern America shows that as the population was increasing, agricultural production was also increasing (Leonard, 1989). According to Djoumes (2021), agricultural production in Africa is mainly increased by an increase in traditional factors such as labour, land, and livestock. Where an increase in labour comes from population growth.

3.11 Summary

This chapter has provided important information related to the study. First, the chapter defined key concepts and provided detailed information on the nature of agricultural production in South Africa. In addition, the chapter provided an overview of public spending on agriculture in South Africa. The chapter also explored two theories of public expenditure, namely Keynesian theory and Wagnerian theory. Moreover, the chapter provided details on how public expenditure ensures agricultural development. Another section focused on the variables that influence the value of agricultural production. Finally, the chapter provided a handful of studies of the national and international context. Most of the studies used the Vector Error Correction Model (VECM) in their empirical analysis. Most of the studies reviewed concerned agricultural production is very limited both in the national and international context.

CHAPTER 4 RESEARCH METHODOLOGY

4.1 Introduction

This chapter provides an overview of the study area and the type of data used in this study. In addition, this chapter provides details on the research design and data collection. Finally, the chapter provides background information on all analysis techniques, diagnostic tests, and variance decomposition used in this study.

4.2 Study area

South Africa is in the southern part of Africa. South Africa is bordered by Namibia, Botswana, Zimbabwe, Mozambique, and Swaziland as shown in Figure 4.1. The agricultural sector is counted among the most important sectors in the country. The South African agricultural sector is known for producing sufficient products which form part of the exports. Hence agricultural output contributes significantly towards economic growth. In South Africa, the main manufacturing sectors are food processing and the production of textiles, metals, and chemicals (Mabin *et al.*, 2022).



Figure 4.1: South African map

Source: Infoplease (2021)

4.3 Data type

This study used publicly available South African time series data for the period 1983 to 2019. The data consist of 37 years of observation in the value of agricultural production, government expenditure in agriculture, annual average rainfall, consumer price index, food import value, and population. The data on the value of agricultural production was aggregated since the agricultural sector is divided into smallholder agriculture and commercial agriculture.

4.4 Data collection

The data were collected from the Food and Agricultural Organisation (FAO), South African Reserve Bank (SARB), the World Bank, and Quantec databases. This study used the value of agricultural production as the dependent variable. The independent variables used are government expenditure in agriculture, average annual rainfall, consumer price index, food import value, and population. Table 4.1 summarises all variables used for analysis in the study and their sources.

Variables	Indicator	Measurement	Source
VAP	Value of agricultural	Million rand	Quantec
	production		
GEA	Government expenditure in	Million rand SARB	
	agriculture		
AAP	Annual average rainfall	Millimetre	World Bank
CPI	Consumer price index	Annual%	World Bank
FIV	Food import value	Million rand	Quantec
PG	Population	Number	FAO

Table 4.1: Description and source of study variables

4.5 Analytical techniques

Econometric Views (EViews) 12 student version statistical tool was used to analyse the data. EViews is an econometric software package for use with time series, crosssectional or longitudinal (panel) data. EViews is used to manage data, perform econometric and statistical analysis, create forecasts or model simulations, and produce high-quality graphics and charts for publication or inclusion in other applications (EViews, 2019).

In achieving all empirical results, the study used multiple econometric procedures. The Augmented Dicky-Fuller (ADF) method was the first technique in examining the stationarity of the selected variables as mandatory for time series data. The second procedure included examining the cointegration relationship between the variables through using the Johansen cointegration test. The third procedure determined the causality between government spending in agriculture and the value of agricultural production using the Granger causality test. The Vector autoregression (VAR) model was used in investigating the relationship among selected variables. For diagnostic purposes, the study used autocorrelation, stability test, heteroscedasticity test, and normality test. The study also used impulse response analysis and variance decomposition. The different techniques are elaborated in the sub-sections that follow.

4.5.1 Unit root testing

Unit root testing aims to test for the stationarity of time series data. Unit root testing is crucial for time series data since the results acquired with non-stationary time series can only be used for that certain period and cannot be used for predicting future values (Banda *et al.*, 2014). This study used the Augmented Dickey-Fuller (1979) method to test for the existence or non-existence of unit roots in the variables that were used in this study, namely, government spending in agriculture, average annual rainfall, consumer price index, food import value, and population. According to Shafuda (2015), the Augmented Dickey-Fuller method tests for a unit root and ultimately control higher-order serial correlation in a time series.

The Augmented Dickey-Fuller test can be conducted using the following regression (Dickey and Fuller, 1979):

Where ΔY_t is the first difference of the series Y, μ_i is a stochastic error term, in which $\Delta Y_{t-1} = (Y_{t-1} - Y_{t-2}), \Delta Y_{t-2} = (Y_{t-2} - Y_{t-3})$. B₁ is a constant, t is the time, β and ϑ are parameters.

The unit root specification model for this study can be expressed as follows:

$\Delta VAP = zVAP_{t-1} + \mu_i \dots \dots$	1.1)
$\Delta GEA = zGEA_{t-1} + \mu_i \dots \dots$.2)
$\Delta CPI = zCPI_{t-1} + \mu_i \dots \dots$.3)
$\Delta AAR = zAAR_{t-1} + \mu_i \dots \dots$.4)
$\Delta FIV = zFIV_{t-1} + \mu_i \dots \dots$.5)
$\Delta PG = zPG_{t-1} + \mu_i \dots \dots$.6)

Where VAP denotes the value of agricultural production, GEA stands for Government expenditure in agriculture, CPI consumer price index, AAR annual average rainfall, FIV food import value, and PG population.

4.5.2 The choice of lag length for VAR

This section elaborates on the selection of the maximum lag that a model should be based on. There are different lag criteria in estimating the Vector autoregressive model. There is the sequential modified LR test statistic acceptable at a 5% level of significance. Other lag testing criteria include Final prediction error, Akaike information criterion, Schwarz information criterion, and Hannan-Quinn information criterion. The Schwarz information criterion and the Akaike information criterion are commonly used while the Hannan-Quinn information criterion is also used but not applicable in most cases.

Mathematical expressions of different lag selection criteria as stated by Mahlangu (2019) are as follows:

Where $\hat{\Sigma}_{\mu}$ is the variance-covariance of the residual, n^{*} is the total number of parameters, *K'* is the total number of regressors in all equations, and T is the sample size.

4.5.3 Cointegration testing

The study used the Johansen test of cointegration to address the first objective of analysing the long-run and short-run relationship between government spending in agriculture and the value of agricultural production. A Johansen cointegration test is a vector autoregressive-based cointegration test (Johansen, 1988). Johansen proposes two different likelihood ratio tests which are named the trace test and the maximum eigenvalue test.

$J_{trace} = -T \sum_{i=r+1}^{n}$	$\ln(1-\lambda_i)$	 	 	

Where T is the sample size, λ_i is the ith largest canonical correlation and r is the number of cointegrating vectors.

4.5.4 Causality testing

The Granger causality test was used to address objective 2 of determining the causality between government spending in agriculture and the value of agricultural production. The Granger causality test is used to test for causality between subsets of variables. Moreover, the Granger causality test establishes the direction of causality among the variables that are included in the system and establish the relevance of one variable in predicting future trends of other variables (Rasheed and Tahir, 2012). The general formula employed by Granger (1969) is given as:

Where X_i and Y_i are variables included in the system. Equations 8 and 9 imply that X_i is granger causing Y_t when a_j is not zero. Additionally, Y_t granger causes X_i when c_j is not zero.

The null hypothesis for the bivariate equations 8 and 9 is that X_i does not granger cause Y_i in the first regression and Y_i does not granger cause X_i in the second regression for all pairs of (x, y) series according to (Granger and Newbold, 1974).

4.5.5 Vector autoregression technique

Vector autoregressive (VAR) is a model based on the statistical properties of available data. The VAR was chosen for this study for the idea of building each model according to all the endogenous variables of the system. VAR provides a consistent and credible approach to data description, structural inference, forecasting, and policy analysis. The model is typically used in forecasting interconnected time series systems (Gou, 2017).

Where t = 1, ..., T, Y_t denotes a k vector of endogenous variables and X_t represents a d vector of exogenous variables. A_i and B are the matrices of the coefficients to be estimated.

The Vector autoregressive model was used to achieve objective 3 of determining the relationship between the value of agricultural production and selected independent

variables: government spending in agriculture (GEA), annual average rainfall (AAR), consumer price index (CPI), food import value (FIV), and population growth (PG). To achieve one outcome from multiple variables, the Vector autoregressive model is the most suitable technique (Endaylalu, 2019).

The specific model that was used for data analysis for this study is stated as follows:

In equation 10.1 the value of agricultural production is expressed as the function of GEA, CPI, AAR, FIV, and PG. Therefore, it is necessary to transform equation 10.1 into a multiple linear regression model as follows:

To make the relationship between the value of agricultural production and independent variables more linear and clearer, the log transformation is introduced to equation 10.2 and forms the new model as equation 10.3.

 $lnVAP = \beta_0 + \beta_1 lnGEA + \beta_2 lnCPI + \beta_3 lnAAR + \beta_4 lnFIV + \beta_5 lnPG + \mu..... (10.3)$ Where:

VAP = Value of agricultural production

GSA = Government spending in agriculture

AAR= Average annual rainfall

CPI = Consumer price index

FIV = Food import value

PG= Population

4.6 Diagnostic tests

This study used diagnostic tests such as autocorrelation test, stability test, heteroscedasticity test, and normality test to check whether the residual of the series passes all of them. Diagnostic tests are used to make earlier predictions on whether the model-building process is valid or not. According to Safi and Al-Reqep (2014), diagnostic checking in time series model building is a significant stage.

4.6.1 Autocorrelation

Autocorrelation or serial correlation is the cross-correlation of a signal with itself at different points in time. According to Hamilton (1994), the most common problem when using time series data is autocorrelation which is characterised by correlated estimated residuals. Autocorrelation simply means the similarities among observations as a function of the time lag among them. This study used the Residual Portmanteau test for autocorrelations which is a model for Q-Statistics for testing serial correlation (Mahdi, 2016). The Portmanteau test for autocorrelations was initially introduced by Box and Pierce (1970). Although the test has undergone many transformations over the years, it is still the most used test for autocorrelation. When the Portmanteau test is employed in its simple form, it is the sample size multiplied by the sum of p-squared sample autocorrelations which can be compared to critical values of a chi-squared distribution ith certain degrees of freedom (Zhu *et al.*, 2017).

The most common Portmanteau test formula based on the residual autocorrelation coefficient is provided as follows according to Peña and Rodríguez (2006):

Where $\hat{\varepsilon}_1, \ldots, \hat{\varepsilon}_t$, are the residuals obtained after estimating the model in a sample of size n, and \hat{r}_k is the residual autocorrelation coefficient.

4.6.2 Stability test

This study used the AR Roots graph to test for stability, in which the graph reports the inverse roots of the characteristic AR polynomial. A Vector autoregressive (VAR) model is concluded to be stable if all the roots lie inside the circle. The AR Roots also help in analysing the impulse response, if the VAR model is not stable then the results for the impulse response may not be valid. According to Milanzi (2012), the Autoregressive (AR) graph shows whether the system is stationary or not. The main aim for conducting a stability test is to establish whether a model is stable or not.

4.6.3 Heteroscedasticity

Heteroscedasticity refers to a situation in which the variance of the regression error term or disturbance term conditional on the independent variables is not constant, (Stock and Watson, 2012). This study used the heteroscedasticity test to establish whether there is an existence of heteroscedasticity problem in the residuals or not. In

addition, the heteroscedasticity test was conducted on individual components and on joint basis for more reliable results. The test is subject to the null and alternative hypotheses as follows:

H₀: There is no heteroscedasticity

H1: There is heteroscedasticity

The decision rule is that the null hypothesis is rejected when the p-value is less than the 0.05 level of significance.

4.6.4 Normality test

The normality test examines whether the residuals are normally distributed or not. This study used the VAR Residual Normality Tests, Cholesky (Lutkepohl) which produced three normality test results which are Skewness, Kurtosis, and Jarque-Bera to check whether the residuals are normally distributed or not. Skewness, Kurtosis, and Jarque-Bera to check Bera coefficients are given as follows according to (Jarque and Bera, 1987)

Where Y1 and Y2 are skewness and kurtosis coefficients respectively.

For a given data set of Y1, Yn, the estimates of μ_2 , μ_3 , and μ_4 are obtained from the sample moments, in which the jth sample moment is given as:

$$m_j = \frac{1}{n} \sum_{t=1}^{n} (Y_t - \bar{Y}) j.....(14)$$

Estimates of Y1 and Y2 are then obtained as:

$$Y1 = \frac{m_3}{(\sqrt{m_2})^3}.....(12.1)$$

Where JB is the Jarque-Bera coefficient, N is the number of observations, K is the number of predictor variables, Y1 is the skewness of the sample's distribution, and Y2 is the kurtosis of the sample's distribution.

The decision rule is that if the P-value of all the tests is less than 0.05 (5%) significance level, then the null hypothesis that the residuals are normally distributed should be rejected and the alternative hypothesis that states that residuals are not normally distributed should be accepted (Hossain, 2015).

4.7 Impulse response analysis

Impulse response analysis is usually conducted to determine and establish how dependent variables respond to a shock in the error term directed to one or several equations included in the VAR model. Impulse response analysis makes it easy to understand the estimated coefficients in the VAR model (Ronayne, 2011). According to Singla and Beag (2014), Granger causality tests usually fail to show recurring feedback from variables in the analysis, resulting in granger causality results being insufficient in research. As a result of the predicament, impulse response analysis is considered the most appropriate way of interpreting models while considering transmission patterns and causality (Killian, 2001). Impulse response functions are used in determining and tracking the reactions of a system's variables to impulses of the system's shock. This study employed the impulse response analysis to consider the period of variables after the exogenous shocks.

4.8 Variance decomposition

According to Brooks (2008), variance decompositions examine the VAR model characteristics differently and thus provide significant information about the relative significance of each random innovation in affecting the VAR variables. Variance decomposition establishes a proportion of dependent variable movements which are a result of their shocks and the shocks of other variables included in the VAR model (Gujarati, 2004). Variance decomposition gives a marginal technique in analysing the VAR system elements. Variance decompositions offer the length of variations in the response of variables associated with the shocks of other variables in the system. This study adopted variance decomposition to understand how much each variable contributes to other variables in the VAR system.

4.9 Summary

This chapter gave brief information on the study area. The chapter also gave information on the nature of data and where the data was sourced from. Selected variables that are used in this study include the value of agricultural production, government expenditure in agriculture, consumer price index, average annual rainfall, food import value, and population. These variables were selected because they are important in influencing the value of agricultural production.

This chapter also gave an overview of econometric techniques used in the analysis. The study used the Augmented Dickey-Fuller Test to test for unit root existence, Johansen cointegration to test for a long-run relationship among the variables, and Vector autoregressive model (VAR) to analyse the relationship among the variables. The VAR model was chosen because of its ability to treat all variables as dependent variables. Hence enabling the study to broaden the analysis to how selected variables are influenced by each other. This chapter also gave an overview of diagnostic tests that the study used. The study used autocorrelation tests, stability tests, heteroscedasticity, and normality tests. Additionally, the chapter included an explanation of impulse response analysis which the study used to check how each variable acts towards the shocks transmitted by other selected variables. Lastly, the chapter included variance decomposition which the study used to check the length of variations in the response of variables associated with the shocks of other variables.

The following chapter presents all the results obtained through using techniques that are mentioned in this chapter to fulfil the objectives of the study.

CHAPTER 5 RESULTS AND DISCUSSION

5.1 Introduction

Chapter 5 shows and describes the details of the research methodology through empirical results that were obtained using econometric procedures. This chapter shows and discusses results that were obtained in descriptive statistics; Augmented Dickey-Fuller test (ADF); Johansen cointegration test; Granger causality test; Vector autoregressive model (VAR); diagnostic tests which include autocorrelation, stability test, heteroscedasticity, and normality test. Furthermore, the chapter shows and discusses the results of the impulse response functions and variance decomposition.

5.2. Descriptive statistics

This study begins with analysing the descriptive statistics of all variables because it is crucial for any econometric analysis.

	VAP	GEA	AAR	CPI	FIV	PG	
	(Million	(Million	(Millimetres)	(Annual%)	Million	(number)	
	Rand)	Rand)			Rand		
Mean	91391.64	8570.432	451.79	69.299	26823.08	45083.78	
Median	52185.60	5190.00	435.70	63.314	10704.40	45571.27	
Maximum	286095.9	22289.00	624.40	158.927	92203.90	58558.27	
Minimum	7881.40	833.00	314.20	9.616	1073.90	30993.76	
Std. Dev.	86601.48	7291.56	75.11	43.95	30449.67	8035092	
Skewness	1.01	0.71	0.43	0.44	1.06	-0.086	
Kurtosis	2.75	1.92	2.46	2.13	2.66	1.929	
Jarque-Bera	6.47	4.91	1.61	2.39	7.17	1.811	
Probability	0.039	0.085	0.445	0.302	0.027	0.404	
Sum	3381491	317106	16716.50	2564.07	992453.9	1668100	
Sum Sq. Dev.	2.70E+11	1.91E+09	203113.7	69550.24	3.34E+10	2.32E+02	
Observations	37	37	37	37	37	37	
Where VAP is the value of agricultural production, GEA is government expenditure in							
agriculture, AAR is annual average rainfall, CPI is a consumer price index, FIV is food import							
value and PG is	population.						

Table 5.1: Descriptive statistics of the time series data

Source: Own computation (2021)

Table 5.1 shows detailed information regarding the time series data that was used to generate all the results in this study. The information that is included in Table 5.1 includes mean, median, maximum, minimum, standard deviation, skewness, kurtosis, Jarque-Bera, probability, sum, the sum of squared deviation, and the number of observations. Table 5.1 shows that the maximum VAP in the data set was R286095.9 and the minimum was R7881.48. Additionally, the average VAP was R52185.60. The maximum GEA was R22289.00, while the minimum was R833.00, and the average GEA was R8570.432. The maximum annual rainfall was 624.40 mil, while the minimum was 314.20 mil, and the average was 415.79 mil. CPI had a maximum of R92203.90, while the minimum was R1073.90 and the average were R26823.08. The maximum PG was 58558.27, while the minimum population was 30993.76, and the average population was 45083.78.

5.3. Unit root results

Usually, time-series data is characterised by a stochastic trend which can be removed by differencing the data. Unit root tests determine whether time series data is stationary or non-stationary and the results of the analysis are presented in Figure 5.1.



Figure 5.1: Graphical illustrations of variables in levels for the period of 1983 to 2019

Source: Own computation (2021)

According to figure 5.1, The value of agricultural production (VAP), Government expenditure in agriculture (GEA), Consumer price index (CPI), Food import value (FIV), and Population (PG) portray a growing trend throughout 1983 to 2019. While Average annual rainfall (AAR) has a fluctuating trend, the data for average annual rainfall (AAR) fluctuates whereas in other years the average rainfall increases then suddenly decrease in the following year. The data for the value of agricultural production (VAP), Government expenditure in agriculture (GEA), Consumer price



index (CPI), Food import value (FIV), and Population (PG) has a trending nature showing that the time series must be differenced for it to be stationary.

Figure 5.2: Graphical illustrations of variables in first difference for a period of 1983 to 2019

Source: Own computation (2021)

Figure 5.2 illustrates the first difference form of the value of agricultural production (VAP), Government expenditure in agriculture (GEA), Consumer price index (CPI), Average annual rainfall (AAR), Food import value (FIV), and Population (PG). Figure 5.2 is different from figure 5.1 in a manner that figure 5.2 shows the stationarity of all the selected variables. The time-series data is integrated of order I(1) to ensure stationarity and to avoid spurious regression. For more reliability, a formal test using the Augmented Dickey-Fuller test was conducted and results presented in Table 5.2:

Variables	Formula	ADF				
		Levels	5% critical	1 st	5% critical	
			value	difference	value	
VAP	Intercept	-3.111**	-2.951	-2.222	-2.976	
	Trend and	-0.706	-3.587	-0.898	-3.587	
	intercept					
	None	3.760**	1.951	3.169**	-1.593	
GEA	Intercept	0.969	-2.945	-6.393**	2.948	
	Trend and	-1.434	3.540-	-6.881**	-3.544	
	intercept					
	none	3.057**	-1.950	-0.852	-1.952	
AAR	Intercept	-5.507**	-2.945	-4.554**	-2.957	
	Trend and	-5.777**	-3.540	-4.326**	-3.595	
	intercept					
	None	-0.477	-1.951	-4.616**	-1.951	
CPI	Intercept	3.069**	-2.951	-2.472	-2.948	
	Trend and	-0.754	-3.548	-4.034**	-3.548	
	intercept					
	None	-2.817**	-1.951	1.080	-1.952	
FIV	Intercept	-1.749	-2.971	-1.271	-2.957	
	Trend and	-2.427	-3.580	-5.892**	-3.548	
	intercept					
	None	-1.575	-1.952	-0.474	-1.951	
PG	Intercept	-1.030	-2.957	-3.277**	-2.967	
	Trend and	-6.878**	-3.552	-2.651	-3.557	
	intercept					
	None	1.409	-1.951	-0.518	-1.951	

Table 5.2: Augmented Dickey-Fuller test results for stationarity

Notes: Reported values under levels and first difference are ADF t-statistics values

* Statistically significant at 1% level

** Statistically significant at 5% level

*** Statistically significant at 10% level

Source: Own computation (2021)

Table 5.2 shows the results acquired through the Augmented Dickey-Fuller test. The results show that the value of agricultural production (VAP), Government expenditure in agriculture (GEA), Consumer price index (CPI), Average annual rainfall (AAR), and Population (PG) were stationary at levels on either intercept, trend, and intercept, or none. However, the Food import value (FIV) was not stationary at levels on all equations. The variables were differenced at levels to reject the null hypothesis that states that there is a unit root existing in the time series data. After differencing, all the variables were found to be stationary. Therefore, the variables are integrated of order I(1), and the null hypothesis of unit root existence is rejected.

5.4 Lag order selection criteria

Lag order selection is an important step in model building. This study used the information criteria approach to select the correct lag. Lag selection in this study is important in cointegration analysis, estimation of the Vector autoregressive model (VAR), Granger causality test, and an impulse response analysis.

Lag	LogL	LR	FPE	AIC	SC	HQ		
0	-2019.344	NA	7.38e+42	115.7340	116.0006	115.8260		
1	-1794.744	359.3606	1.59e+38	104.9568	106.8232	105.6011		
2	-1739.838	69.02402*	6.54e+37*	103.8765*	107.3427	105.0730*		
* indi	cates lag order se	elected by th	e criterion					
(each	test at 5% level)							
LR: s	equentially modif	ied LR test s	tatistic					
FPE: Final prediction error								
AIC: Akaike information criterion								
SC: S	SC: Schwarz information criterion							
HQ: F	Hannan-Quinn inf	ormation crit	erion					

Table 5.3: Lag order selection criteria

Source: Own computation (2021)

The results in Table 5.3 show that lag length 2 is suggested by 4 criteria which are LR, FPE, AIC, and HQ. The choice of lag length can affect the results of cointegration and VAR, hence in this case it was convenient to choose the most optimal lag order which is 2.

5.5 Johansen cointegration results

Since the Augmented Dickey-Fuller test (ADF) showed stationarity of selected variables, it is important to determine the long-run relationship that exists among the variables. In this study, the Johansen cointegration test was conducted to obtain the long-run relationship results. The Johansen cointegration test provides the number of cointegrating equations. Furthermore, this study adopted the two tests namely the Trace test and the Maximum eigenvalue test. This study determines the long-run relationship between the value of agricultural production and other selected variables, then separately between the value of agricultural production and government expenditure alone.

		Trace test		Maximum eigenvalue	
				test	
Hypothesised	Eigenvalue	Trace	0.05	Maximum	0.05
no. of CE(s)		statistics	critical	eigenvalue	critical
			value	statistics	value
None*	0938	192.076**	95.753	55.004**	40.077
At most 1*	0.708	97.072**	69.818	41.966**	33.876
At most 2*	0.551	55.106**	47.856	27.268	27.584
At most 3	0.424	27.838	29.797	18.807	21.131
At most 4	0.226	9.030	15.494	8.711	14.264
At most 5	0.009	0.319	3.841	0.319	3.841

Trace test indicates 3 cointegrating eqn(s) at the 0.05 level

Max-eigenvalue test indicates 2 cointegrating equation(s) at the 0.05 level

* Denotes rejection of the hypothesis at the 0.05 level

Source: Own computation (2021)

Table 5.4 illustrates the results of the Johansen cointegration test when all the variables are included. From the trace test, there are three cointegrating equations at a 5% level of significance. From the maximum eigenvalue test, there are 2 cointegrating equations at a 5% level of significance. The rule of thumb states that the null hypothesis is rejected when there is no cointegration among the variables. Therefore, the null hypothesis of no cointegrating equations for the trace test is

rejected at none because the trace statistic is 192.076 which is greater than the critical value of 95.004 at a 5% level of significance. Additionally, the null hypothesis is also rejected for the trace test at (at most 1) because the trace statistic is 97.072 which is greater than the critical value of 69.818 at a 5% significance level. Moreover, the null hypothesis is rejected at (at most 2) since the trace statistic is 55.106 which is greater than the critical value of 47.856. Therefore, based on the trace test it is concluded that there are cointegrating equations confirming that there is a long-run relationship that exists between the variables.

The maximum eigenvalue test shows Maximum eigenvalue statistics of 55.004 at (none), which is greater than the critical value of 40.077. There is a cointegrating equation at none based on these findings. In addition, the maximum eigenvalue test shows a maximum eigenvalue statistic of 41.996 at (at most 1) which is greater than the critical value of 33.876 at a 5% level of significance. Therefore, there is a cointegrating equation at (at most 1). The null hypothesis that there is no cointegration is rejected based on the maximum eigenvalue test, confirming that there is a long-run relationship between the variables.

		Trace test		Maximum eigenvalue			
				test			
Hypothesised	Eigenvalue	Trace	0.05	Maximum	0.05		
no. of CE(s)		statistics	critical	eigenvalue	critical		
			value	statistics	value		
None*	0.465245	21.614**	15.494	21.282**	14.284		
At most 1	0.009735	0.332	3.841	0.332	3.841		
Trace test indica	ates 1 cointeg	rating eqn(s)	at the 0.05 lev	/el			
Max-eigenvalue test indicates 1 cointegrating equation(s) at the 0.05 level							
* denotes rejection of the hypothesis at the 0.05 level							
Source: Own cor	nputation (202	21)					

Table 5.5: Johansen Cointegration test results for VAP and GEA only

Table 5.5 shows the Johansen cointegration test results for the value of agricultural production and government spending in agriculture. Based on the trace test, there is a cointegrating equation at (none) because the trace statistic of 21.614 is greater than

the critical value of 15.494. Therefore, the null hypothesis that there is no cointegration is rejected based on the trace test. Additionally, the null hypothesis that there is no cointegration is rejected based on the maximum eigenvalue test because the maximum eigenvalue statistic is 21.282 at (none), which is greater than the critical value of 14.284. Therefore, it is concluded that there is a long-run relationship between the value of agricultural production and government expenditure in agriculture in South Africa (The nature of this relationship is explained in subsection 5.7.2.2). These findings are in line with the findings of Chandio et al. (2016) who analysed the impact of government expenditure on the agricultural sector and economic growth in Pakistan. Chandio et al. (2016) used the Johansen cointegration test to check for the long-run relationship between government expenditure in agriculture and agricultural production output. The results of Chandio et al. (2016) confirmed that there is a longrun relationship that exists between government expenditure in agriculture and agricultural production output based on both the trace test and the maximum eigenvalue test. Additionally, these results concur with the findings of Umair et al. (2017) who analysed government expenditure and some representative agricultural product prices and their impact on the agricultural growth of Pakistan. Umair et al. (2017) used an Autoregressive Distributed Lag (ARDL) model and established that there is a long-run relationship that exists between government expenditure in agriculture and agricultural economic growth and the relationship is manifested through the prices of agricultural products.

5.6 Granger causality results

The Johansen cointegration test proved that there is cointegration between the value of agricultural production and other selected variables. This section checks the causality using the Granger causality test. The Granger causality test gives results of the causal association among selected variables, and it gives the directional causality between two variables. Table 5.6 shows the causal association between the value of agricultural production and all selected variables based on the probability values.

Table 5.6: Granger Causality results

Null Hypothesis	Obs.	F-Stat.	Prob.	Decision
DGEA does not Granger cause DVAP	34 included	0.24541	0.7840	Accept
DVAP does not Granger cause DGEA		1.59284	0.2206	Accept
DCPI does not Granger cause DVAP	34 included	0.39003	0.6805	Accept
DVAP does not Granger cause DCPI		2.08565	0.1425	Accept
DAAR does not granger cause DVAP	34 included	0.42173	0.6599	Accept
DVAP does not Granger cause DAAR		2.13975	0.1359	Accept
DFIV does not Granger cause DVAP	34 included	9.92527	0.0005**	Reject
DVAP does not Granger cause DFIV		1.46711	0.2472	Accept
DPG does not Granger cause DVAP	34 included	0.54354	0.5865	Accept
DVAP does not Granger cause DPG		0.29980	0.7432	Accept
Notes: Granger cause if P < 0.05				
* Statistically significant at 1% level				
** Statistically significant at 5% level				
*** Statistically significant at 10% level				

Source: Own computation (2021)

The results shows that government expenditure in agriculture does not granger cause the value of agricultural production. These results are in line with the results of Oinam *et al.* (2022) who analysed the impact of public expenditure on agricultural growth in India. Oinam *et al.* (2020) used a Pairwise Granger causality test and concluded that government expenditure in agriculture in India does not granger cause agricultural GDP because government expenditure is not optimally allocated in the agricultural sector. In South Africa, government expenditure in agriculture is still poorly allocated (Aguera *et al.*, 2020) and would therefore not stimulate substantial increases in agricultural output. Which is the possible reason why government expenditure in agriculture in agricultural production just like in India, contemplating that poor governance in resource allocation can lead to failure in addressing economic objectives such as the objectives of the fiscal policy (Nguyeng and Luong, 2021).

These results do not concur with the study of Mile *et al.* (2021) which made an empirical analysis of government agricultural spending and agricultural output value in

Nigeria. The study of Mile *et al.* (2021) used the Vector Error Correction Granger Causality/Block homogeneity Wald test and found that government agricultural spending granger causes gross agricultural output value because Nigeria allocates 5% of its total government expenditure to the agricultural sector. Unlike in South Africa where the government only allocate 1% of its total government expenditure to the agricultural sector as stated in the introduction chapter. Therefore, both Oinman *et al.* (2022) and Mile *et al.* (2021) give emphasis on the reasons why government expenditure in agriculture does not granger cause the value of agricultural production in South Africa. These results imply that government expenditure in agriculture alone is not useful in forecasting future values of the value of agricultural production. However, it does not mean that there is no relationship between the value of agricultural production and government expenditure in agriculture (The Johansen Cointegration test results established a long-run relationship existence between the two variables).

Additionally, the study findings show that the value of agricultural production does not granger cause government expenditure in agriculture. These findings are in line with the study of Bassey and Amaraihu (2018) which was conducted in Nigeria titled agricultural expenditure, Maputo declaration target and agricultural output: A case study of Nigeria. Bassey and Amaraihu (2018) used a Pairwise Granger Causality test to check the causality between actual agricultural expenditure and agricultural output and established that agricultural output in Nigeria does not granger cause government expenditure. Even though the Johansen cointegration test proved that there is a long-run relationship between the value of agricultural production and government expenditure in agriculture, the Granger causality test proved that there is no causality between the two variables. Hence these findings prove that Wagner's theory does not hold for the agricultural sector in South Africa because Wagner (1876) as defined in the literature review chapter states that the causality between economic growth and government expenditure runs from economic growth to government expenditure and not vice versa.

In other results, the only variable that granger causes the value of agricultural production from all selected variables is food import value. The generated probability from the Granger causality test is 0.0005 which is less than the 5% level of significance. Therefore, implying that food import value granger causes the value of

49

agricultural production in South Africa. These results shows that food import value is crucial in forecasting future values of the value of agricultural production (a further discussion of this relationship is elaborated in 5.7.1.1). These results are not in line with the study of Syed (2015) which analysed the impact of agricultural imports and exports on agricultural productivity in Pakistan. Syed (2015) used the pairwise granger causality test to check for causal association between food imports and agricultural productivity, the study found that food imports do not granger cause agricultural productivity in Pakistan.

5.7 Vector autoregressive model results

A Vector autoregressive (VAR) is an n-equation, n-variable model that is linear. In a VAR model, each variable is explained by its own lagged values and past or current values of the other included n-1 variables. Table 5.7 presents the results of the VAR model.

	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
InVAP (-1)	0.638779	-0.267	0.061	-0.889	-0.167	0.002
	(0.211)	(0.560)	(0.048)	(0.520)	(0.5480)	(0.001)
	[3.017]**	[-0.477]	[1.280]	[-1.710]	[-1.710]	[1.895]
InVAP (-2)	-0.073	0.576	-0.064	0.332	-0.282	0.004
	(0.167)	(0.443)	(0.038)	(0.412)	(0.434)	(0.001)
	[-0.435]	[1.297]	[-1.694]	[0.807]	[-0.649]	[3.414] **
InGEA (-1)	0.161	0.748	0.031	-0.360	0.194	-0.001
	(0.077)	(0.205)	(0.017)	(0.191)	(0.201)	(0.000)
	[2.082]**	[3.639]**	[1.757]	[-1.887]	[0.963]	[3.166] **
InGEA (-2)	-0.033	-0.137	-0.014	0.498	0.297	-0.000
	(0.069)	(0.183)	(0.015)	(0.170)	(0.179)	(0.000)
	[-0.488]	[-0.750]	[-0.886]	[2.927]**	[1.653]	[-1.487]
InCPI (-1)	-0.208	1.150	0.988	0.478	-0.119	-0.003
	(0.617)	(1.633)	(0.140)	(1.516)	(1.600)	(0.004)
	[-0.337]	[0.704]	[7.013]**	[0.315]	[-0.074]	[-0.733]
InCPI (-2)	-0.564	-1.589	-0.273	0.165	-0.859	-0.004

 Table 5.7: Vector Autoregressive model results

	(0.625)	(1.654)	(0.142)	(1.536)	(1.620)	(0.004)
	[-0.903]	[-0.960]	[-1.919]	[0.107]	[-0.530]	[1.024]
InAAR (-1)	0.088	-0.235	0.033	-0.119	0.229	0.000
	(0.066)	(0.174)	(0.015)	(0.162)	(0.171)	(0.000)
	[1.346]	[-1.351]	[2.226]**	[-0.739]	[1.340]	[1.246]
InAAR (-2)	0.120	0.083	0.059	-0.108	0.007	0.001
	(0.073)	(0.193)	(0.016)	(0.180)	(0.189)	(0.000)
	[1.639]	[0.431]	[3.561]**	[-0.601]	[0.041]	[2.209] **
InFIV (-1)	0.146	0.481	-0.023	0.252	0.354	-0.000
	(0.070)	(0.185)	(0.015)	(0.172)	(0.181)	(0.000)
	[2.097]**	[2.601]**	[-1.478]	[1.465]	[1.952]	[-0.441]
InFIV (-2)	-0.234	-0.517	-0.043	0.318	-0.199	-0.002
	(0.076)	(0.202)	(0.017)	(0.188)	(0.198)	(0.000)
	[-3.066]**	[-2.558]**	[-2.502]**	[1.695]	[-1.006]	[-3.735]**
InPG (-1)	4.397	-6.393	5.838	-35.667	11.595	1.822
	(6.377)	(16.871)	(1.455)	(15.671)	(16.531)	(0.046)
	[0.689]	[-0.378]	[4.010]**	[-2.275]**	[0.701]	[38.903]**
InPG (-2)	1.527	8.699	-4.175	29.769	-0.373	-0.860
	(5.951)	(15.745)	(1.358)	(14.625)	(15.428)	(0.043)
	[0.256]	[0.552]	[-3.073]**	[2.035]**	[-0.024]	-19.683
С	-57.095	-21.567	-16.738	68.242	-109.044	-0.325
	(17.904)	(47.368)	(4.087)	(43.999)	(46.414)	(0.131)
	[-3.188]	[-0.455]	[-4.095]	[1.550]	[-2.349]	[2.473]
R squared	0.997	0.982	0.999	0.548	0.992	0.999
Adj. R	0.996	0.972	0.999	0.302	0.988	0.999
squared						
Sum aq.	0.069	0.485	0.003	0.419	0.466	3.75E-06
Resids						
S.E	0.056	0.148	0.012	0.138	0.145	0.000
equation						
F.statistic	906.805	101.016	8613.787	2.228	252.859	473379.8

Log-	59.246	25.195	110.949	27.777	25.907	231.213
likelihood						
Akaike	-2.642	-0.696	-5.597	-0.844	-0.737	-12.469
AIC						
Schwarz	-2.064	-0.119	-5.019	-0.266	-0.159	-11.891
Mean	11.029	8.743	4.077	6.108	9.495	10.719
dependent						
No parentheses, (.), [.] represents the coefficient, standard error, and t-stat						
respectively.						
** represents statistical significance at 95%						

Source: Own computation (2021)

5.7.1 Discussion summary of study findings presented in Table 5.7

5.7.1.1 Government expenditure in agriculture

The results in Table 5.7 show that there is a significant positive relationship between the one-year lagged government expenditure in agriculture and government expenditure in agriculture. A 1% increase in a one-year lagged government expenditure increases government expenditure in agriculture by 74.8%. The findings are in line with the study of Okpara (2017) which analysed the effect of government expenditure on agriculture and agricultural output on the Nigerian economic growth for the period 1980 to 2015. The study found that government expenditure on agriculture has a positive relationship with government expenditure on agriculture. Additionally, the results show that there is a positive significant relationship between a one-year lagged food import value and government expenditure in agriculture by 48%. However, the results also show that there is a negative relationship between a two-year lagged food import value and government expenditure in agriculture where a 1% increase in food import value and government expenditure in agriculture where a 1% increase in food import value and government expenditure in agriculture where a 1% increase in food import value and government expenditure in agriculture where a 1% increase in food import value and government expenditure in agriculture where a 1% increase in food import value and government expenditure in agriculture where a 1% increase in food import value on a two-year lagged basis reduces government expenditure in agriculture by 51.7%.

5.7.1.2 Consumer Price Index

The results show that there is a positive significant relationship between a one-year lagged consumer price index and the consumer price index. When the consumer price

index increases by 1%, the consumer price index increases by 98.8%. Additionally, the results in Table 5.7 show that there is a positive significant relationship between a one-year lagged average annual rainfall and the consumer price index where a 1% increase in average annual rainfall increases consumer price index by 3.3%. Moreover, the results show that there is a positive significant relationship between a two-year lagged average annual rainfall and the consumer price index where a 1% increase in average annual rainfall and the consumer price index where a 1% increase in average annual rainfall and the consumer price index where a 1% increase in average annual rainfall of the consumer price index where a 1% increase in average annual rainfall increases the consumer price index by 5.9%. These findings are in line with the findings of Octoviani *et al.* (2011) which were based on the impact of global climate change on the Indonesian economy, the findings revealed that a decrease in rainfall reduces agricultural production. Furthermore, the findings state that when production is low the prices of agricultural goods are likely to be high which pushes the consumer price index to be high.

5.7.1.3 Average Annual rainfall

The findings also show that there is a positive significant relationship between a twoyear lagged government expenditure in agriculture and average annual rainfall, whereby an increase in government expenditure in agriculture increases average annual rainfall. The relationship between government expenditure and rainfall might not be direct. Leppanen et al. (2015) analysed the impact of climate change on regional government expenditures in Russia. From the evidence that Leppanen et al. (2015) gathered, a decline in rainfall results from a decrease in government expenditure but the relationship between government expenditure and rainfall is not linear. Jin et al. (2021) assessed how government expenditure mitigates emissions that reduce rainfall. Jin et al. (2021) found that spending on research and development influences the cuts in carbon dioxide levels in both industrialised and emerging economies. Carbon dioxide emissions and rainfall have a negative correlation as stipulated in a study on variations in atmospheric carbon dioxide and its association with rainfall and vegetation over India by Tiwari et al. (2013). Therefore, carbon dioxide concentration is reduced when the government spends on agricultural research and development which indirectly leads to improved rainfall patterns.

Table 5.7 showed the Vector autoregressive model results for all the data series included in the study. For this study, the area of interest was to focus on the first column which showed the lagged independent variables, and the second column

53

which showed the relationship between independent variables with the value of agricultural production (InVAP). However, relying on Table 5.7 alone to emphasize the relationship between the value of agricultural production and selected independent variables was not efficient because the results lack probability values. Therefore, the next step was to generate probability values and sample out the area of focus for this study as shown in Table 5.8. Table 5.8 shows the outsourced VAR model results which are the results that emphasize the relationship between the value of agricultural production and selected independent variables.

InVAP as the dependent variable					
Independent	Coefficient	Std. Error	t-stat	Prob.	
variables					
InVAP(-1)	0.638779	0.211706	3.017297	0.0031**	
InVAP(-2)	-0.073059	0.167798	-0.435402	0.6640	
InGEA(-1)	0.161955	0.077764	2.082641	0.0392**	
InGEA(-2)	-0.033866	0.069297	-0.488704	0.6259	
InCPI(-1)	-0.208038	0.617256	-0.337037	0.7366	
InCPI(-2)	-0.564957	0.625236	-0.903591	0.3679	
InAAR(-1)	0.088903	0.066007	1.346888	0.1803	
InAAR(-2)	0.120143	0.073278	1.639548	0.1035	
InFIV(-1)	0.146890	0.070018	2.097902	0.0378**	
InFIV(-2)	-0.234572	0.076503	-3.066171	0.0026**	
InPG(-1)	4.397032	6.377028	0.689511	0.4917	
InPG(-2)	1.527991	5.951629	0.256735	0.7978	
С	-57.09542	17.90459	-3.188870	0.0018	
Observations: 3	5			·	
R-squared	0.997982	2 Mean de	ependent var	11.02903	
Adjusted R-squared 0.996882		S.D. dependent var		1.005709	
S.E. of regression 0.056160 Sum squared resid. 0.			0.069386		
Durbin-Watson stat 2.418308					

 Table 5.8: Outsourced Vector Autoregression model results

Source: Own computation (2021)

5.7.2 Discussion of statistically significant study findings presented in Table 5.8

Table 5.8 shows the results of the VAR model where the value of agricultural production is the dependent variable. The adjusted R-squared is 99.68% showing that the variation in the value of agricultural production is explained by the included independent variables. The model has no autocorrelation since the Durbin-Watson statistic is 2.41.

5.7.2.1 Value of agricultural production

There is a positive statistically significant relationship between the value of agricultural production and the one-year lagged value of agricultural production. The coefficient of the one-year lagged value of agricultural production is 0.638. Implying that a 1% increase in the value of agricultural production increases the country's value of agricultural production by 63.87% in the current year. The results are consistent with the findings of Iganiga and Unemhilin (2011) who analysed the impact of government agricultural expenditure on the value of agricultural output in Nigeria. Iganiga and Unemhilin (2011) concluded that the value of agricultural output increases the value of agricultural production and the two-year lagged value of agricultural production is negative and statistically insignificant. The coefficient of the two-year lagged value of agricultural production is-0.073. Implying that a 1% increase in the value of agricultural production by 7.30% in the second year.

5.7.2.2 Government expenditure in agriculture

The findings also show a significant positive relationship between the value of agricultural production and the one-year lagged government expenditure in agriculture. A 1% increase in government expenditure in agriculture increases the value of agricultural production by 16.19% in the current year. Implying that increasing government expenditure in agriculture is good for the value of agricultural production. The results are consistent with the study of Ebenezer *et al.* (2019) who analysed the impact of government expenditure in South Africa. Ebenezer *et al.* (2019) concluded that there is a positive long-run relationship between government expenditure in agriculture and agricultural productivity. A negative insignificant relationship between the value of agricultural production and a two-year lagged government expenditure in agriculture in agriculture was also found. Where a 1% increase in government expenditure in agriculture results in a 3.33% decrease in the value of agricultural production. These

55

findings concur with the findings of Selvaraj (1993) who analysed the impact of government expenditure on agriculture and the performance of the agricultural sector in India. Selvaraj (1993) found that government expenditure in agriculture due to its instability is detrimental to agricultural growth in India. These results imply that government expenditure in agriculture increases the value of agricultural in the short-run and negatively affects the value of agricultural production in the long run, entailing the problem of diminishing marginal productivity of government expenditure in the agricultural sector (IMF, 2022).

5.7.2.3 Food import value

There is a positive significant relationship between one-year lagged food import value and the value of agricultural production. A 1% increase in food import value increases the value of agricultural production by 14.68% in the current year. Additionally, the results show that there is a negative significant relationship between the two-year lagged food import value and the value of agricultural production. Where a 1% increase in food import value decreases the value of agricultural production by 23.45%. These findings concur with the findings of Iganiga and Unemhilin (2011) who analysed the impact of government agricultural expenditure on the value of agricultural output in Nigeria. The study of Iganiga and Unemhilin (2011) found that a two-year lagged food import value reduces the value of agricultural production in Nigeria. These results imply that importing food from other countries positively benefits the local agricultural sector in the short-run. However, importing food reduces local food production in the long-run

5.7.3 Discussion of statistically insignificant study findings presented in Table 5.8

5.7.3.1 Consumer Price Index

There is a negative insignificant relationship between the value of agricultural production and a one-year lagged consumer price index. When a one-year lagged consumer price index increases by 1%, the South African value of agricultural production declines by 20.8%. Additionally, Table 5.8 shows that there is a negative insignificant relationship between a two-year lagged consumer price index and the value of agricultural production. Whereby a 1% increase of a two-year lagged consumer price index reduces the value of agricultural production by 56.49%. The results are in line with the findings of Gou (2017) who analysed the relationship

between agricultural product price fluctuation and consumer price index in China. Gou (2017) concluded that an increase in the consumer price index reduces agricultural product prices. Agricultural product prices can be considered as the value of agricultural production.

5.7.3.2 Annual Average rainfall

The results show that there is a positive insignificant relationship between a one-year lagged annual average rainfall and the value of agricultural production. In which a 1% increase in a one-year lagged annual average rainfall leads to an 8.89% increment of the value of agricultural production. Moreover, there is a positive insignificant relationship between a two-year lagged annual average rainfall and the value of agricultural production. A 1% increase in annual average rainfall increases the value of agricultural production by 12.01%. The results are in line with the results of Akinniran *et al.* (2013) who analysed the effect of rainfall variability on crops production in Oyo State, Nigeria. The study of Akinniran *et al.* (2013) concluded that an increase in rainfall increases agricultural production. An increase in agricultural production subsequently increases the value of agricultural production of agricultural production.

5.7.3.3 Population

The results in Table 5.8 show that there is a positive insignificant relationship between a one-year lagged population and the value of agricultural production. A 1% increase in a one-year lagged population increases the value of agricultural production by 4.39%. Additionally, the results show that there is a positive insignificant relationship between a two-year lagged population and the value of agricultural production whereby a 1% increase in population increases the value of agricultural production by 1.52%. These results are in line with the record of Boserup (1965) which highlighted that population increases agricultural output. Moreover, the results are consistent with the study of Schneider *et al.* (2021) which focused on the impacts of population growth, economic development, and technical change on global food production and consumption. Schneider *et al.* (2021) found that population increases global food production.

5.8 Diagnostic test results

This study used diagnostic tests such as autocorrelation test, stability test, heteroscedasticity test, and normality test to check if the model is correctly specified. The VAR model is correctly specified when it passes all the diagnostic tests that are presented in this sub-section.

5.8.1 Autocorrelation

The diagnostic portmanteau test for autocorrelation was introduced by Box and Pierce (1970) based on the asymptotic distribution of the residual autocorrelations. This study used the Residual Portmanteau test for autocorrelation as stated in the methodology chapter. The results of the Residual Portmanteau test are presented in Table 5.9.

Lags	Q-stat	Prob.	Adj. Q-Stat	Prob
1	32.41422	0.6399	33.36758	0.5944
2	67.28969	0.6352	70.35672	0.5328
3	99.21200	0.7154	105.2717	0.5564
4	141.1634	0.5513	152.6362	0.2952
5	165.2629	0.7774	180.7523	0.4702
6	201.9447	0.7451	225.0234	0.3227
7	219.0135	0.9343	246.3594	0.5884
8	242.5161	0.9760	276.8258	0.6714
9	258.8741	0.9886	312.3077	0.6696
10	298.2603	0.9923	353.4484	0.6874
11	317.0195	0.9986	380.8055	0.6996
12	348.1986	0.9988	428.2483	0.5419

Table 5.9: Residual	Portmanteau test	for autocorrelations
---------------------	------------------	----------------------

Source: Own Computation (2021)

The results suggest that there is no autocorrelation problem since the p-values of the Q statistics and adjusted Q statistics are greater than the 0.05 level of significance, the test was made at a 5% level of significance. Furthermore, the results suggest that the first order of the VAR is significant up to the 12th lag.



Figure 5.3: Inverse roots of the AR characteristic polynomial

Source: Own Computation (2021)

Figure 5.3 illustrates the inverse roots of the Autoregressive (AR) characteristic polynomial, in which most inverse roots are within the borders of the circle and one on the circle line. These results imply that the Vector Autoregressive model is stable. Furthermore, the modulus for the roots is less than 1 which shows that the system is stationary.

5.8.3 Heteroscedasticity

The heteroscedasticity test was used to check the existence of inconsistency of the variance to avoid bias and poor performance of the VAR model. When there is no existence of heteroscedasticity then the VAR model is unbiased and produces robust estimations.
5.8.3.1 Test on individual components

Dependent	R-squared	F (24,10)	Prob.	Chi-sq (24)	Prob.				
Res1*res1	0.550876	0.511066	0.9136	19.28067	0.7369				
res2*res2	0.896603	3.613121	0.0194	31.38112	0.1431				
res3*res3	0.632086	0.715843	0.7596	22.12299	0.5719				
res4*res4	0.853664	2.430666	0.0724	29.87825	0.1888				
res5*res5	0.758568	1.309149	0.3391	26.54989	0.3259				
res6*res6	0.750843	1.255641	0.3668	26.27951	0.3391				
res2*res1	0.798966	1.655952	0.2049	27.96381	0.2616				
res3*res1	0.624747	0.693694	0.7780	21.86613	0.5873				
res3*res2	0.739617	1.183538	0.4074	25.88658	0.3589				
res4*res1	0.725130	1.099203	0.4602	25.37956	0.3854				
res4*res2	0.588160	0.595052	0.8563	20.58558	0.6630				
res4*res3	0.841590	2.213636	0.0955	29.45565	0.2035				
res5*res1	0.843626	2.247884	0.0914	29.52691	0.2009				
res5*res2	0.903417	3.897421	0.0148	31.61960	0.1367				
res5*res3	0.803871	1.707784	0.1903	28.13548	0.2544				
res5*res4	0.774908	1.434428	0.2823	27.12178	0.2988				
res6*res1	0.636002	0.728029	0.7494	22.26008	0.5637				
res6*res2	0.474485	0.376206	0.9758	16.60696	0.8649				
res6*res3	0.561363	0.533246	0.8996	19.64771	0.7166				
res6*res4	0.674095	0.861823	0.6373	23.59331	0.4850				
res6*res5	0.523912	0.458522	0.9428	18.33692	0.7864				
* Denotes no ł	* Denotes no heteroscedasticity: Prob.>0.05								

Table 5.10: Heteroscedasticity test results on individual components

Source: Own Computation (2021)

Table 5.10 shows the results of the Heteroscedasticity test on individual components. The results validate that there is no heteroscedasticity problem since all the probability values are greater than the 0.05 level of significance, the test was made at a 5% level of significance.

5.8.3.2 Joint Heteroscedasticity test

Table 5.11: Joint Heteroscedasticity test

Test	Ho	df	T-statistic	Prob.	Conclusion
Chi-sq.	No	504	525.1318	0.2491	Accept H ₀ , Prob.
	heteroscedasticity				Is greater than
					0.05. Therefore,
					there is no
					heteroscedasticity
Accept H	₀ : if Prob. >0.05				

Source: Own computation (2021)

5.8.4 Normality test

The normality test establishes the probability distribution of residuals which is one of the assumptions of the Ordinary Least Squares (OLS) method. The estimates of a model are best linear and unbiased if the residuals follow the normal distribution with zero mean and constant variance. This study used VAR Residual Normality Tests, Cholesky (Lutkepohl) and results are presented in Table 5.12.

Table 5.12: VAR Residual Normality Tests, Cholesky (Lutkepohl)

Component	Skewness	Chi-sq	df	Prob.*
1	0.681677	2.710656	1	0.0997
2	0.439981	1.129235	1	0.2879
3	-0.026239	0.004016	1	0.9495
4	0.354811	0.734363	1	0.3915
5	0.300509	0.526782	1	0.4680
6	0.235003	0.322153	1	0.5703
Joint		5.427206	6	0.4903
Component	Kurtosis	Chi-sq	df	Prob.
1	2.529783	0.322443	1	0.5701
2	3.268545	0.105170	1	0.7457
3	2.466333	0.415335	1	0.5193
4	2.530958	0.320835	1	0.5711
5	4.046593	1.597395	1	0.2063
6	2.489641	0.379847	1	0.5377

Joint	3.141024	6	0.7910
Component	Jarque-Bera	df	Prob.
1	3.033099	2	0.2195
2	1.234405	2	0.5395
3	0.419351	2	0.8108
4	1.055198	2	0.5900
5	2.124177	2	0.3457
6	0.702001	2	0.7040
Joint	8.568230	12	0.7393

Table 5.12 shows the vector autoregressive residual normality tests based on the Cholesky (Lutkepohl) technique which results in three normality tests called Skewness, Kurtosis, and Jarque-Bera tests. This study included all three tests for the viability of results. The null hypothesis is that the residuals are normally distributed, and the alternative hypothesis is that residuals are not normally distributed.

Based on Table 5.11, the results show that the Skewness test statistic is 5.42720 and the P-value is 0.4903 which is greater than 0.05 (5%). Additionally, the Kurtosis test statistic is 3.141024 and the P-value is 0.7910 which is greater than 0.05 (5%). Finally, the Jarque-Bera test statistic is 8.568230 with a P-value of 0.7393 which is greater than 0.05 (5%). Therefore, the null hypothesis is accepted that the residuals are normally distributed since the P-values of all the tests are greater than 0.05 (5%). This decision is derived from the decision rule stated in the methodology chapter.

5.9 Impulse response analysis

The Vector autoregressive (VAR) system allows the analysis of the impulse response of endogenous variables to shocks of other variables in the same system. Impulse response analysis was used to demonstrate the unitary shock on each selected variable. Impulse response analysis traces the effects of a one-time shock to one or more responses of independent variables. Through impulse response analysis, the transmission that occurs between one variable to another can be assessed (Rizwan *et al.,* 2017) and the findings of this study are presented in Figure 5.4 to 5.9. The positive effect is indicated by the middle line above (0.0), and the negative effect is indicated by the middle line above (0.0).



Figure 5.4: Impulse responses to the value of agricultural production

The value of agricultural production (VAP) transmits positive shocks to the value of agricultural production, government expenditure in agriculture (GEA), consumer price index (CPI), and population (PG) for the period of 10 years, while average annual rainfall (AAR) and food import value (FIV) receives both positive and negative shocks. VAP responds positively to the shocks of itself, and this is in line with what the study recorded in Table 5.8 have shown. Even though VAP transmits negative and positive shocks to AAR and PG respectively, there might be no economic sense derived from the relation.



Figure 5.5: Impulse responses to government expenditure in agriculture

Government expenditure in agriculture (GEA) transmits positive shocks to GEA, value of agricultural production (VAP), consumer price index (CPI), food import value (FIV), while average annual rainfall (AAR) receives both positive and negative shocks. However, as stated in the discussion of the VAR model results, the relationship between GEA and AAR is not direct. The result that GEA transmits positive shocks to VAP is in line with what the study recorded in tables 5.7 and 5.8.



Figure 5.6: Impulse responses to the consumer price index

The consumer price index (CPI) transmits positive shocks to CPI for the period of 10 years. The value of agricultural production (VAP), government expenditure in agriculture (GEA), average annual rainfall (AAR), food import value (FIV), and population (PG) receive both positive and negative shocks from CPI in the period of 10 years. Both VAP and GEA respond negatively fast to the shocks of CPI as compared to AAR, FIV, and PG since they start responding negatively from year 2.



Figure 5.7: Impulse responses to average annual rainfall

Average annual rainfall (AAR) transmits positive shocks to consumer price index (CPI) and population (PG) for the 10 years period. AAR, value of agricultural production (VAP), government expenditure in agriculture (GEA), and food import value (FIV) receive both negative and positive shocks. In the VAR model result discussion, it was stated that the relationship between rainfall and CPI is not direct. Instead, rainfall

affects agricultural production and agricultural production directly affects the consumer price index.



Figure 5.8: Impulse responses to food import value

Source: Own computation (2021)

From figure 5.8, it is evident that VAP, GEA, CPI, and FIV receive both positive and negative shocks of FIV while CPI and PG receives only negative shocks for the whole period of 10 years. FIV responds positively to itself in the short-run with a downward

slopping curve. However, FIV also responds negatively to itself in the long-run. VAP receives positive and negative shocks from food import value both in the short-run. GEA responds positively to the shocks of FIV in the short-run, but negatively in the long-run.



Figure 5.9: Impulse responses to population

Source: Own Computation (2021)

Population (PG) transmits positive shocks to the value of agricultural production (VAP), government expenditure in agriculture (GEA), consumer price index (CPI), food import value (FIV), and PG for the period of 10 years. Average annual rainfall (AAR) receives both positive and negative shocks of PG. Even though the relationship

between AAR and PG might not be direct, AAR responds negatively to the shocks of PG. However, AAR starts responding positively in the 7th year.

5.10 Variance decomposition

Variance decomposition provides details on the amount that each variable contributes to other variables in the autoregression system. Additionally, variance decomposition specifies how much of the forecast error variance of each variable can be explained by the shocks of other variables. Simply, variance decomposition determines how much a change in endogenous variables is going to the fluctuation in the exogenous variable over time. The interpretation in this area of variance decomposition is based on period three in the short-run and period four to ten in the long run. Cholesky order of variables has been used in this results presentation. The tables given under this test show the aggregate results of such decomposition.

Period	S. E	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
1	0.056	100.000	0.000	0.000	0.000	0.000	0.000
2	0.085	77.445	18.117	0.138	0.345	3.931	0.021
3	0.096	65.136	26.948	0.324	4.435	3.074	0.080
4	0.100	60.341	29.702	2.256	4.308	3.266	0.124
5	0.104	56.448	31.013	3.996	5.331	3.050	0.159
6	0.108	53.978	31.826	4.443	6.435	3.099	0.218
7	0.113	53.541	32.559	4.640	6.019	2.901	0.336
8	0.116	52.877	32.878	5.132	5.774	2.844	0.491
9	0.118	52.279	32.819	5.831	5.549	2.881	0.638
10	0.121	52.189	32.624	6.335	5.306	2.781	0.762

 Table 5.13: Variance decomposition of InVAP

Source: Own Computation (2021)

The results in Table 5.13 indicate that shocks of the value of agricultural production (VAP) were the main drivers of VAP in South Africa as compared to other selected variables. Therefore, these findings mean that the South African value of agricultural production in the context of the VAR system used in the study can be predicted by its previous behaviour. After the first year, the main influence on South Africa's VAP was government expenditure in agriculture shocks (InGEA, accounting 18.11%), followed

by food import value (InFIV, accounting 3.93%), consumer price index (InCPI, accounting 0.13%), and average annual rainfall (InAAR, accounting 0.34%). The magnitude of population contribution to VAP did not change significantly between the 2nd year and the 10th year (InPG, contributing 0.02% to 0.76%), meaning that the population did not have a significant effect on the value of agricultural production. These findings are in line with the findings of the VAR model recorded in Table 5.8.

Period	S. E	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
1	0.148	4.055	95.944	0.000	0.000	0.000	0.000
2	0.214	2.632	83.800	0.992	5.852	6.714	0.007
3	0.233	5.085	82.851	0.862	5.221	5.953	0.026
4	0.241	5.222	81.651	1.178	5.101	6.778	0.067
5	0.245	5.672	79.532	2.036	5.016	7.630	0.110
6	0.248	6.840	77.541	2.873	4.905	7.697	0.140
7	0.252	8.688	75.368	3.490	4.767	7.517	0.167
8	0.2551	9.916	73.844	4.014	4.677	7.359	0.188
9	0.257	10.475	72.923	4.534	4.609	7.253	0.204
10	0.258	10.742	72.294	4.975	4.596	7.176	0.215

Table 5.14: Variance decomposition of InGEA

Source: Own Computation (2021)

According to Table 5.14, the variation of government expenditure in agriculture (GEA) is explained mostly by the shocks of GEA over the period of 10 years (accounting for almost 96% to 72% from year 1 to year 10 respectively). These findings mean that the South African government expenditure in agriculture can be predicted by its previous behaviour. The main driver of government expenditure in agriculture from the 2nd year was the value of agricultural production (InVAP, accounting for 2.63% to 10.74% in the 10th year), followed by food import value (InFIV, accounting for 6.71% in the second year to 7% in the 10th year), and average annual rainfall (InAAR, accounting 5.85%). Consumer price index shocks accounted for 0.9% in the second year. However, the magnitude of CPI shocks to GEA increase from 0.99% to almost 5% on the 10th year, meaning that consumer price index in South Africa played a big role in the reduction of government expenditure in line with what the study recorded in Table

5.7. Population shocks accounted the least in the variation of GEA as compared to other variables for the whole period of 10 years.

Period	S. E	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
1	0.012	5.691	0.184	94.123	0.000	0.000	0.000
2	0.021	17.350	4.827	69.040	6.596	1.603	0.581
3	0.030	15.097	9.292	48.167	22.436	3.683	1.322
4	0.035	12.742	8.419	40.384	29.697	6.754	2.000
5	0.038	13.946	8.254	36.343	30.699	8.169	2.585
6	0.041	19.481	10.027	32.208	28.176	7.207	2.900
7	0.046	25.547	13.270	27.026	25.315	5.910	2.928
8	0.050	28.800	16.827	22.701	23.653	5.120	2.896
9	0.054	30.531	20.081	19.619	22.182	4.717	2.868
10	0.058	32.150	22.872	17.352	20.448	4.353	2.822

Table 5.15: Variance decomposition of InCPI

Source: Own Computation (2021)

The findings in Table 5.15 shows that the main driver of consumer price index (CPI) in the short-run (1st and 2nd year) is consumer price (InCPI, accounting 94%). However, the main drivers of CPI from the 8th year were the value of agricultural production (InVAP, accounting 29%) and average annual rainfall (InAAR, accounting almost 24%). The effect of AAR shocks on CPI supports what the study recorded in section 5.7.1.2, where it was discussed that AAR increases CPI but the relationship between the two variables is indirect. The shocks of government expenditure (GEA) on CPI increased from 0.18% in the 1st year to 22.87% in the tenth year. Therefore, these findings shows that the main drivers of the South African consumer price index in the long-run contemplating the study period were VAP, GEA, and ARR and the least drivers were food import value (FIV) and population (PG). Meaning that the shocks of FIV and PG were not significant on the variation of CPI.

Period	S.E	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
1	0.138	9.749	3.156	1.488	85.605	0.000	0.000
2	0.163	19.823	11.088	1.163	64.321	3.216	0.386
3	0.164	19.560	11.757	1.171	63.503	3.412	0.594
4	0.182	16.909	14.593	2.033	56.278	9.549	0.636
5	0.185	16.905	16.112	2.510	54.442	9.384	0.644
6	0.188	18.908	15.882	2.497	52.841	9.238	0.631
7	0.189	19.815	15.679	2.472	52.080	9.329	0.622
8	0.190	19.830	15.906	2.515	51.829	9.298	0.618
9	0.191	19.741	16.173	2.603	51.585	9.279	0.617
10	0.191	19.623	16.364	2.667	51.464	9.264	0.615

Table 5.16: Variance decomposition of InAAR

The findings shows that the main driver of average annual rainfall (AAR) in South Africa is AAR. Other factors that influenced ARR mostly in South Africa in the study period include the value of agricultural production (InVAP, accounting 9.74% in the 1st year and 19.62% in the 10th year) and government expenditure in agriculture (InGEA, accounting 3% in the first year and 16% in the 10th year). These findings support what the study recorded in section 5.7.1.3 where it was discussed that the relationship between government expenditure in agriculture and average annual rainfall is not direct. Furthermore, these findings shows that as much as government expenditure in agriculture in South Africa is considered a driver of average annual rainfall, the relationship between the two variables is bilateral because it was discussed under Table 5.14 that average annual rainfall is one of the drivers of government expenditure in agriculture.

Period	S. E	InVAP	InGEA	InCPI	InAAR	InFIV
1	0.145	6.245	18.971	2.890	9.339	62.553
2	0.164	6.329	27.767	2.862	8.019	54.980
3	0.180	7.113	37.550	2.475	6.708	46.069
4	0.193	6.894	43.096	2.239	7.507	40.142

 Table 5.17: Variance decomposition of InFIV

InPG

0.000

0.039

0.082

0.120

5	0.201	6.890	45.599	2.057	8.307	36.950	0.194
6	0.208	9.281	46.187	1.920	7.747	34.524	0.338
7	0.215	11.383	45.240	1.906	7.874	33.086	0.509
8	0.219	12.988	43.880	2.130	8.078	32.269	0.653
9	0.224	14.897	42.706	2.416	7.937	31.288	0.752
10	0.228	17.165	41.700	2.609	7.672	30.035	0.816

Based on Table 5.17, the variation in food import value (FIV) is mainly explained by FIV in the period of 10 years. These findings mean that FIV in South Africa can be explained by its own past values. Main drivers of FIV in South Africa from the 1st year were shocks of government expenditure in agriculture (InGEA, accounting 19%), average annual rainfall (InAAR, accounting 9%), and the value of agricultural production (InVAP, accounting 6%). Population (PG) shocks accounted less in the variation of FIV as compared to other selected variables accounting 0.03% and 0.8% in 2nd year and 10th year respectively. In addition, the magnitude of population shocks did not increase drastically for the period of 10 years, meaning that population is not significant in predicting FIV in line with what the study recorded in Table 5.7 of the VAR model results.

Period	S.E	InVAP	InGEA	InCPI	InAAR	InFIV	InPG
1	0.000	4.447	35.048	12.035	1.078	0.018	47.370
2	0.001	13.363	48.228	6.037	2.492	0.116	29.762
3	0.002	25.205	45.977	4.122	4.414	0.476	19.804
4	0.003	34.677	42.409	2.984	5.423	0.758	13.746
5	0.004	40.524	40.089	2.112	6.317	0.947	10.008
6	0.006	43.672	38.984	1.448	6.979	1.100	7.814
7	0.007	45.478	38.570	0.975	7.275	1.233	6.467
8	0.009	46.691	38.578	0.705	7.157	1.293	5.574
9	0.010	47.592	38.784	0.628	6.776	1.280	4.938
10	0.011	48.227	39.066	0.726	6.283	1.229	4.466

Source: Own Computation (2021)

According to Table 5.18, the variation of the population is mainly explained by population (PG) in the short-run (2 years) and by the value of agricultural production (VAP) and government expenditure in agriculture (GEA) in the long-run (10 years). Meaning that PG can be predicted by its own past values only in the short-run. However, PG is mainly influenced by VAP and GEA in the long-run, meaning that the shocks of VAP and GEA contribute significantly to the variation of the South African population even though the contribution might not be direct. CPI shocks accounted for a proportion of 12.03% in the first year, the proportion declined sharply to 0.72% in the tenth year. AAR had a proportion of 0.07% in the first year which increased to 6.38% in the tenth year. Finally, FIV shocks accounted for a proportion of 0.01% in the first year which increased to 1.22% by the tenth year.

5.11 Summary

This chapter presented and discussed the econometric results of the study. Additional information on the data used as well as some analysis tables are provided in the Appendix. This chapter established 9 sections in which the first one gave descriptive statistics of the data set. The second section gave stationarity tests using the Augmented Dickey-Fuller test. The data was found to be stationary at first difference as indicated by the results in Table 5.7. For the VAR model implementation, two lags were chosen using the automatic lag selection criteria shown in Table 5.3. Section four used the Johansen cointegration test to test for the long-run relationship and found that 3 equations were cointegrating as indicated in Table 5.4. Furthermore, results for cointegration on the value of agricultural production and government expenditure showed that there is a long-run relationship between the two variables as indicated in Table 4.5.

The fifth section gave results of the Granger causality test in which food import value was found to Granger cause the value of agricultural production. The chapter also gave results of the Vector autoregressive model (VAR) as shown in Table 5.7. Table 5.8 further gave results of the VAR model only focusing on the value of agricultural production as the dependent variable. The results showed that the value of agricultural production, government expenditure in agriculture, and food import value is positive in determining the value of agricultural production and they are statistically significant.

74

Other variables that are positive towards the value of agricultural production include average annual rainfall and population.

The chapter also gave the results of diagnostic tests. Autocorrelation was tested using the residual portmanteau test and no autocorrelation was found to exist in the data. The chapter also gave results of the stability test using the Inverse Roots of the Autoregressive characteristic polynomial and the model was found to be stable. Heteroscedasticity was tested on individual residuals and presented in Table 5.9, and on joint basis represented in Table 5.10. Both results on heteroscedasticity showed that no heteroscedasticity problem exists on the data. The chapter also gave results for normality based on Skewness, Kurtosis, and Jarque-Bera models. All the tests for normality showed that the residuals are normally distributed. The chapter gave results of the impulse analysis for all variables. Lastly, the chapter gave results of the variance decomposition of all variables.

CHAPTER 6

SUMMARY, CONCLUSION, AND POLICY RECOMMENDATIONS

6.1 Introduction

This chapter summarises the main findings of the study. Additionally, the chapter gives conclusions based on the findings and gives policy recommendations. Furthermore, the chapter gives suggestions on areas that can be researched in the future. Lastly, the chapter gives an overview of the limitations of this study.

6.2 Summary of findings

The study aimed to examine the relationship between the value of agricultural production, government spending in agriculture, annual average rainfall, consumer price index, food import value, and population. The first objective was to analyse the short-run and long-run relationship between government spending in agriculture and the value of agricultural production. The study used the Johansen cointegration test to check the long-run relationship between the two variables. The study found that the value of agricultural production and government expenditure in agriculture have a long-run relationship. The study used the Vector autoregressive (VAR) model to check the short-run relationship between the value of agricultural production and government expenditure in agriculture. The estimated model found that a one-year lagged government expenditure in agriculture in agriculture has a positive significant relationship with the value of agricultural production.

The second objective was to determine the causality between government spending in agriculture and the value of agricultural production. The study used the Granger causality test to address this objective. The results of the Granger causality test showed that government expenditure in agriculture does not granger cause the value of agricultural production and the value of agricultural production does not granger cause government expenditure in agriculture.

The third objective was to determine the relationship between the value of agricultural production and selected independent variables: government spending in agriculture, annual average rainfall, consumer price index, food import value, and population. The study used the VAR model to address this objective.

76

The study found that the one-year lagged value of agricultural production, one-year lagged government expenditure in agriculture, and one-year lagged food import value have a positive significant relationship with the value of agricultural production. Additionally, the one-year lagged value of agricultural production and one-year lagged government expenditure in agriculture have the greatest positive significant effect on the value of agricultural production, contributing 63.87% and 16.19% respectively. Even though a one-year lagged food import value was found to have a positive significant relationship with the value of agricultural production, the two-year lagged food import value was found to have a negative significant relationship with the value of agricultural production.

In both one-year lag and two-year lag basis, the study found that consumer price index, average annual rainfall, and population have an insignificant relationship with the value of agricultural production. Even though the one-year lagged consumer price index reduces the value of agricultural production by 20.45% and the two-year lagged consumer price index reduces the value of agricultural production by 50.49%, the consumer price index does not have the greatest negative impact because of its relationship with the value of agricultural production is statistically insignificant. Therefore, the variable that has the greatest negative impact on the value of agricultural production is food import value since its relationship with the value of agricultural production is statistically significant and reduces the value of agricultural production by 23.45% on a two-year lagged basis.

6.3 Conclusions

This study had three hypotheses and they were all tested.

Hypothesis one: The null hypothesis that there is no significant long-run and shortrun relationship between government expenditure in agriculture and the value of agricultural production was rejected because the Johansen cointegration test proved that there is a long-run relationship between the two variables. Furthermore, the VAR model proved that there is a short-run relationship between government expenditure in agriculture and the value of agricultural production.

Hypothesis two: The null hypothesis that there is no causality between government expenditure in agriculture and the value of agricultural production cannot be rejected because the Granger causality test proved that the value of agricultural production

does not granger cause government expenditure in agriculture and government expenditure in agriculture does not granger cause the value of agricultural production.

Hypothesis three: The null hypothesis that there is no relationship between the value of agricultural production and selected independent variables: government spending in agriculture, annual average rainfall, consumer price index, food import value, and the population was rejected because the VAR model results proved that all selected variables have either a positive or negative effect towards the value of agricultural production.

6.4 Recommendations

This study researched the main problem of the fluctuating value of agricultural production as stated in the problem statement. Given the positive influence of government expenditure on the value of agricultural production, it is recommended that the government should increase its expenditure on the agricultural sector. Increasing government expenditure will enable the agricultural sector to produce better quality research and enhance modern efficient technology aimed at increasing agricultural production and stabilizing the value of agricultural production. In addition, the government should increase its spending in the agricultural sector by investing in research related to climate-smart agriculture. Investing in climate-smart research will ensure that climate-friendly technologies are developed and adopted. Climate-smart technologies will reduce carbon dioxide emissions and increase rainfall which is efficient for the agricultural sector as discussed in section 5.7.1.3.

The study found that there is a negative relationship between the consumer price index and the value of agricultural production. An increase in the consumer price index decreases the value of agricultural production. It is recommended that policymakers review the monetary policy of South Africa which controls inflation (measured in the consumer price index) to ensure price stability. This will ensure that prices grow at an average pace rather than drastically changing. Subsequently, the value of agricultural production which is measured as the gross price of agricultural production output at farm-gate will be sustained from fluctuating.

The study found that food import value granger causes the value of agricultural production, and the study also found that there is a negative relationship between the value of agricultural production and food import value. Therefore, it is recommendable

78

that the South African government should discourage imports and encourage South African agricultural producers to produce more major imported food products such as rice, wheat, palm oil, cane sugar, chicken cuts and offal, whiskies, sunflower oil, and soybean meal as reported by the International Trade Administration (2022). Engaging in producing the aforementioned products locally will ensure that South Africa imports less of them and thus lowering the food import value and ultimately stabilising the value of agricultural production.

6.5 Limitations of the study

This study might not have focused on all variables that affect the value of agricultural production due to limited data availability and time. Additionally, the study referenced literature that was on agricultural output and agricultural productivity because the literature related to the value of agricultural production is very limited. However, this meant that this study could add to the knowledge that is limited in terms of analysing the specific variable of the value of agricultural production.

6.6 Suggestions for further studies

To add more literature concerning the value of agricultural production, future research can investigate the effect of other variables such as interest rates, policy reforms, land, temperature, the value of exports, and fuel prices. In addition, future research can focus on modelling technical and allocative efficiencies of government expenditure in agriculture.

REFERENCES

Adewuyi, A., Ogebe, J.O., and Oshota, A. (2021). The Role of Exchange Rate and Relative Import Price on Sawnwood Import Demand in Africa: Evidence from Modified Heterogeneous Panel Data Methods. *The Journal of Economic Asymmetries*, vol. 24, no. 2021, pp. e00231.

AgriSETA. (2016). Annual Report, viewed 29 July 2021, http://www.agriseta.co.za

Aguera, P., Berglund, N., Chimembiri, T., Comninos, A., Gillwald, A., and Govan-Vassen, N. (2020). Paving the Way Towards Digitalizing Agriculture in South Africa. Research ICT South Africa, viewed 19 March 2022, <u>https://researchictafrica.net/wp/wpcontent/uploads/2020/09/Pavingthewaytowardsdig</u> <u>italisingagricultureinSouthAfricaWhitepaper272020105251.pdf</u>

Ahmed, T., Khan, KS., and Naeem, M. (2019). The Effect of public spending on agricultural growth: Evidence from 1972 to 2014 in Pakistan. *Sarhad Journal of Agriculture*, vol. 36, no. 1, pp. 311-318.

Aigheyisi, O. (2011). Examining the Relative Effectiveness of Monetary and Fiscal Policies in Nigeria: A Cointegration and Error Correction Approach. *SSRN Electronic Journal*, viewed 10 September 2021, <u>https://ssrn.com/abstract=1944585</u>

Allen, C., Asmal, Z., Bhorat, H., Hill, R., Monnakgotla, J., Oosthuizen, M., and Rooney, C. (2021). Employment Creation Potential, Labour Skills Requirements, and Skill Gaps for Young People: A South African Case Study. Development Policy Research Unit, working paper no. 202102, viewed 5 April 2022, http://www.dpru.uct.ac.za/Publications/Working_Papers/DPRU%20WP202102.pdf

Altson, J., Beddow, J., and Pardey, P. (2009). Agricultural Research, Productivity, and Food Prices in the Long Run. *Science (New York, N.Y.)*, vol. 325, no. 10, pp. 1209-1210.

Amare, M., Jensen, N.D., Shiferaw, B., and Cissé, J.D. (2021). Rainfall Shocks and Agricultural Productivity: Implication for Rural Household Consumption. *Agricultural Systems*, vol. 166, no. 14, pp. 79-89.

Atayi, A., Boniface, J., Bobola, A., and Olorunrinu, A. (2020). The Effect of Government Expenditure on Agricultural Output in Nigeria (1981-2018). *GPH-International Journal of Applied Management Science*, vol. 3, no.11, pp. 01-14.

Australian Bureau of Agricultural and Resource Economics and Sciences (ABARES). (2021). Agricultural Productivity. *Research topics*, viewed 19 January 2022, <u>https://www.awe.gov.au/abares/research-topics/productivity</u>

Awubare, D. B., and Eyitope, J. A. (2015). The Effects of Public Expenditure on Agricultural Production Output in Nigeria. *Journal of Research in Humanities and Social Science*, vol. 3, no. 11, pp. 07-23.

Ayoub, Z., and Michel, Z. (2019). The Impact of Public Expenditure on the Agricultural Sector Productivity in China. *American Journal of Humanities and Social Sciences Research*, vol. 3, no. 9, pp.173-180.

Baba, S. H., Saini, A. S., Sharma, K. D. and Thakur, D. R. (2010). Impact of Investment on Agricultural Growth and Rural Development in Himachal Pradesh: Dynamics of Public and Private Investment. *Indian Journal of Agricultural Economics*, vol. 65, no. 1, pp. 135-58.

Badar, H., Ghafoor, A., and Sultan., A.A. (2007). Factors Affecting Agricultural Production. *Department of Marketing and Agribusiness*, vol. 44, no. 3, pp. 202-207.

Banda, H., Ngirande, H., and Hogwe, F. (2016). The Impact of Economic Growth on Unemployment. *Investment Management and Financial Innovation*, vol. 13, no. 2, pp. 246-255.

Baranauskaite, L., and Jureviciene, D. (2021). Import Risks of Agricultural Products in Foreign Trade. *Economies*, vol. 9, no. 3, pp. 102-117.

Bassey, E., and Amaraihu, C. (2018). Agricultural Expenditure, Maputo Declaration Target, and Agricultural Output: A Case Study of Nigeria. *International Journal of Economics, Commerce and Management*, vol. 5, no. 7, pp. 516-531.

Bernstein, H. (2013). Commercial Agriculture in South Africa Since 1994: Natural, Simply Capitalism. *Journal of Agrarian Change*, vol. 13, no. 1, pp. 23-46.

Biza, R., Kapingura, F., and Tsegaye, A. (2015). Do Budget Deficits Crowd Out Private Investment? An Analysis of the South African Economy. *International Journal of Economic Policy in Emerging Economies*, vol. 8, no. 10, pp. 204-211.

Black, A., and Gerwel, H. (2014). Shifting the Growth Path to Achieve Employment Intensive Growth in South Africa. *Development Southern Africa*, vol. 31, no. 2, pp. 241-256.

Boserup, E. (1965). The Conditions of Agricultural Growth. United Kingdom (London): George Allen and Unwin Ltd.

Box, G.E.P., and Pierce, D.A. (1970). Distribution of Residual Autocorrelation in Autoregressive-Integrated Moving Average Time Series Models. *Journal of the American Statistical Association*, vol. 65, no. 332, pp. 1509–1526.

Brooks, C. (2008). Introductory Econometrics for Finance (2nd ed.). United States of America (New York): Cambridge University Press.

Brownson, S., Vincent, I.M., Emmanuel, G., and Etim, D. (2012). Agricultural Productivity and Macro-Economic Variable Fluctuation in Nigeria. *International Journal of Economics and Finance*, vol. 4, no. 8, pp. 114-125.

Carter, M.I. (1984). Identification of the Inverse Relationship between Farm Size and Productivity: An Empirical Analysis of Peasant Agricultural Production. *Oxford Economic Papers* New Series, vol. 36, no. 1, pp. 131-145.

Cervantes-Godoy, D., and Dewbre, J. (2010). Economic Importance of Agriculture for Poverty Reduction, OECD Food, Agriculture and Fisheries Working Papers, no. 23, viewed 19 January 2022, <u>https://www.oecd.org/brazil/44804637.pdf</u>

Chandio, A., Yuansheng, J., Rehman, A., and Jingdong, L. (2016). Impact of Government Expenditure on Agricultural Sector and Economic Growth in Pakistan. *International Journal of Biology and Biotechnology*, vol. 7, no. 3, pp. 1046-1053.

Chen, B., Lv, B., and Liu, Y. (2019). Financial Development and the Composition of Government Expenditure: Theory and Cross-Country Evidence. *International Review of Economics and Finance*, vol. 64, no. 59, pp. 600-611.

Chipaumire, G., Ngirande, H., Mangena, M., and Yewukai, R. (2014). The Impact of Government Spending on Economic Growth: Case South Africa. *Mediterranean Journal of Social Sciences*, vol. 5. No. 10, pp. 109-118.

Curristine, T., Lonti, Z., and Joumard I. (2007). Improving Public Sector Efficiency: Challenges and Opportunities. *OECD Journal on Budgeting*, vol. 7, no. 1, pp. 1-41.

Dao, M. (2012). Population and Economic Growth in Developing Countries. *International Journal of Academic Research in Business and the Social Sciences*, vol. 2, no. 1, pp. 6-17.

Department of Agriculture Forestry and Fisheries (DAFF). (2017). Trends in the agricultural sector, viewed 19 February 2022, <u>http://www.daff.co.za</u>.

Department of Agriculture, Forestry and Fisheries (DAFF). (2018). National Policy on Comprehensive Producer Development Support: Draft 5, viewed 19 January 2022, https://www.daff.gov.za/docs/Policy%20on%20CPDS_30May2018_accepted%20cha https://www.daff.gov.za/docs/Policy%20on%20CPDS_30May2018_accepted%20cha https://www.daff.gov.za/docs/Policy%20on%20CPDS_30May2018_accepted%20cha https://www.daff.gov.za/docs/Policy%20on%20CPDS_30May2018_accepted%20cha

Department of Agriculture, Forestry and Fisheries (DAFF). (2019). Economic Review of the South African Agriculture, viewed 04 May 2021, <u>https://www.dalrrd.gov.za</u>

Devarajan, S., Swaroop, V., and Zou, H.F. (1996). The Composition of Public Expenditure and Economic Growth. *Journal of Monetary Economics*, vol. 37, no. 2, pp. 313-344.

Dickey, A.D., and Fuller, A.Y. (1979). Distribution of the Estimators for Autoregressive Time Series with a Unit Root. *Journal of the American Statistical Association*, vol. 74, no. 366, pp. 427-431.

Djokoto, J.G., Gidiglo, F.K., Srofenyoh, F.Y., Henaku, K.A., Prah, C.B., and Arthur, A.A.A. (2021). Agricultural Development in the Presence of Foreign Divestment: Policy Options. *Journal of Agriculture and Food Research*, vol. 71, no. 50, pp. 2666-1543.

Djoumes, Y.F. (2021). What Innovations Impact Agricultural Productivity in Sub-Saharan Africa? *Journal of Agriculture and Food Research*, vol. 6, no. 2021, pp. 10028.

Dong, Y., Guo, B., He, D., Liao, X., Zhang, Z., Wu, X. (2022). Industrial Transformation and Urban Economic Efficiency Evolution: An Empirical Study of the Yangtze River Economic Belt. *International Journal of Environmental Research and Public Health*, vol. 19, no. 4154, pp.

Ebenezer, M., Ngarava, S., Etim, N.A., and Popoola, O.O. (2019). Impact of Government Expenditure on Agricultural Productivity in South Africa. *The Journal of Social Sciences Research*, vol. 5, no. 12, pp. 1734-1742.

Edeh, C.E., Eze, C.G., and Oguanyi, S.O. (2020). Impact of Foreign Direct Investment on the Agricultural Sector in Nigeria (1981-2017). *African development review*, vol. 32, no. 4, pp. 551-564.

Endaylalu, S. (2019). The Impact of Public Expenditure Components on Economic Growth in Ethiopia; Vector Autoregressive Approach. *International Journal of Business and Economics Research*, vol. 8, no. 4, pp. 211-219.

Ernawati, E., Tajuddin, T., and Nur, S. (2021). Does Government Expenditure Affect Regional Inclusive Growth? An Experience of Implementing Village Fund Policy in Indonesia. *Economies*, vol. 9, no. 4, pp. 164-176.

Fan, S., and Rao, N. (2003). Public Spending in Developing Countries: Trends, Determination, and Impact. International Food Policy Research Institute, EPTD Discussion Paper no. 99, viewed 20 February 2022, https://ageconsearch.umn.edu/record/16080/files/ep030099.pdf

Food and Agricultural Organisation (FAO). (2015). Statistic Documents, viewed 06 August 2021, <u>http://www.fao.org/3/i4646e/i4646e.pdf</u>

Food and agricultural organisation (FAO). (2016). Decent Rural Employment, Productivity Effects and Poverty Reduction in Sub-Saharan Africa, Rural Transformations Technical Papers Series 05, viewed 04 May 2021, http://www.fao.org/3/i5432e/i5432e.pdf

Food and Agricultural Organisation (FAO). (2021). Statistic Documents viewed 03 May 2021, <u>http://fenixservices.fao.org/faostat/static/documents/QV/QV_e.pdf</u>

Gao, X., Ji, L., Chandio, A.A., Gul, A., Ankrah-Twumasi, M., and Ahmad, F. (2022). Towards Sustainable Agriculture in China: Assessing the Robust Role of Green Public Investment. Sustainability. *Sustainability*, vol. 14, no. 6, pp. 3613-3630.

Giang, M.H., Xuan, T.D., Trung, B.H., and Que, M.T. (2017). Total Factor Productivity of Agricultural Firms in Vietnam and Its Relevant Determinants. *Economies*, vol. 7, no. 4, pp. 4-16.

Gou, G.H. (2017). The Study on the Relationship between Agricultural Product Price Fluctuation and Inflation. *Journal of Service Science and Management*, vol. 10, no. 15, pp.166-176.

Granger, C., and Newbold, P. (1974). Spurious Regressions in Econometrics. *Journal* of econometrics, vol. 2, no. 2, pp. 111-120.

Granger, C.W.J. (1969). Investigating Causal Relations by Econometric Models and Cross Spectral Methods. *Journal of Econometrica*, vol. 37, no. 3, pp. 424-38.

Greyling, J. 2015. A look at the contribution of the agricultural sector to the South African Economy. Department of agricultural economics, Stellenbosch University, viewed 20 January 2022, <u>https://core.ac.uk/download/pdf/37401071.pdf</u>

Greyling, J., Mabaya, E., and Nick, V. (2015). South Africa's Agricultural Sector Twenty Years After Democracy (1994 to 2013). *Professional Agricultural Workers Journal*, vol. 3, no. 1, pp. 10-24.

Gujarati, D.N. (2004). Basic Econometrics. New York: The McGraw-Hill Companies.

Hamilton, J.D. (1994). Time Series Analysis. Econometrics, viewed 25 January 2022, http://www.ru.ac.bd/stat/wpcontent/uploads/sites/25/2019/03/504_02_Hamilton_Time -Series-Analysis.pdf

Hossain, A. (2015). Vector Autoregressive (VAR) Modelling and Projection of DSE. *Chinese Business Review*, vol. 14, no. 6, pp. 273-289.

Huka, H., Ruoja, C., and Mchopa, A. (2014). Price Fluctuation of Agricultural Products and its Impact on Small Scale Farmers Development: Case Analysis from Kilimanjaro Tanzania. *European Journal of Business and Management*, vol. 6, no. 36, pp.155-160. Hull, S., Babalola, K., and Whittal, J. (2019). Theories of Land Reform and Their Impact on Land Reform Success in Southern Africa. *Land*, vol. 8, no. 11, pp.178-206.

Ibok, O.W., and Bassey, N.E. (2014). Wagner's Law Revisited: The Case of Nigerian Agricultural Sector (1961-2012). *International Journal of Food and Agricultural Economics*, vol. 2, no. 3, pp. 19-32.

Iganiga, B., and Unemhilin, D. (2011). The Impact of Federal Government Agricultural Expenditure on Agricultural Output in Nigeria. *Journal of Economics*, vol. 2, no. 2, pp. 81-88.

Infoplease. (2021). Map of South Africa, viewed 16 September 2021, https://www.infoplease.com/atlas/africa/south-africa-map

International Monetary Fund (IMF). (2021). Fiscal Policy, viewed 20 February 2022, https://www.imf.org/external/pubs/ft/fandd/basics/fiscpol.htm

International Monetary Fund (IMF). 2022. Unproductive Public Expenditures: A Pragmatic Approach to Policy Analysis, viewed 20 February 2022, https://www.imf.org/external/pubs/ft/pam/pam48/pam4801.htm

Iternational Trade Administration (ITA). (2022). South Africa-Country Commercial Guide. Agricultural Sector, viewed 16 April 2022, <u>https://www.trade.gov/country-commercial-guides/south-africa-agricultural-sector</u>

Jambo, N. (2017). The Impact of Government Spending on Agricultural Growth: A Case of Zambia, Malawi, South Africa, and Tanzania. Stellenbosch University thesis, viewed 16 September 2021, https://scholar.sun.ac.za/bitstream/handle/10019.1/101022/jambo_impact_2017.pdf?

Jaramillo, S., Graterol, E., and Pulver, E. (2020). Sustainable Transformation of Rainfed to Irrigated Agriculture Through Water Harvesting and Smart Crop Management Practices. *Front Sustain Food System*, vol. 4, no. 3, pp.70-86.

Jarque, C., and Bera, A. (1987). A Test for Normality of Observations and Regressions Residuals. *International Statistical Review*, vol. 55, no. 2, pp. 163-172.

Jefferson, P., Joliveth, S., and Audre Garett, A. (2022). Relationship Between Government Expenditure on Agriculture and Agricultural Production in the Republic of

the Philippines. College of Commerce of Business Administration, University of Santo Tomas, Manila, Philippines, viewed 05 September 2022, <u>https://www.researchgate.net/publication/361162911_Relationship_Between_Gover_ nment_Expenditure_on_Agriculture_and_Agricultural_Production_in_the_Republic_ of_the_Philippines</u>

Jin, Y., Tang, Y.M., Chau, K.Y., and Abbas, M. (2021). How Government Expenditure Mitigates Emissions: A Step Towards Sustainable Green Economy in Belt and Road initiatives project. *Journal of Environmental Management*, viewed 10 December 2021, https://doi.org/10.1016/j.jenvman.2021.113967

Johansen, S. (1988). Statistical Analysis of Cointegration Vectors. *Journal of Economic Dynamics and Control*, vol. 12, no. 2, pp. 231–254.

Karfakis, P., Velazco, J., Moreno, E., and Covarrubias, K. (2011). Impact of Increasing Prices of Agricultural Commodities on Poverty. ESA Working Paper no. 11-14, viewed 25 January 2022, https://www.fao.org/3/am320e/am320e.pdf

Kargbo, J.M. (2007). The effects of macroeconomic factors on South African agriculture. *Applied Economics*, vol. 39, no. 17, pp. 2211-2230.

Kashif, U., Hong, C., and Naseem, D.R. (2017). Comparison Of Government Expenditure and Some Representative of Agricultural Product Prices and their Impact on Agricultural Growth of Pakistan. *Russian Journal of Agricultural and Socio-Economic Sciences*, vol. 63, no. 3, pp. 79-85.

Keynes, J.M. (1936). The General Theory of Employment, Interest, and Money. Macmillan, United Kingdom (London).

Khapayi, M.I., and Celliers, P.R. (2016). Factors Limiting and Preventing Emerging Farmers to Progress to Commercial Agricultural Farming in the King William's Town area of the Eastern Cape Province, South Africa. *South African Journal of Agricultural Extension*, vol. 44, no. 1, pp. 374-384.

Khashei, M., and Bijari, M. (2012). Hybridization of the Probabilistic Neural Networks with Feed-forward Neural Networks for Forecasting. *Engineering Applications of Artificial Intelligence*, vol. 25, no. 6, pp. 1277-1288.

Kilian, L. (2001). Impulse Response Analysis in Vector Autoregressions with Unknown Lag Order. *Journal of Forecasting*, vol. 20, no. 3, pp. 161-79.

Kirsten, J., & Edwards, L., and Vink, N. (2009). Distortions to Agricultural Incentives in South Africa. Agricultural Distortions Working Paper no. 38, viewed 5 December 2021, <u>https://www.researchgate.net/publication/228355007_Distortions_to_Agricultural_Inc</u> <u>entives_in_South_Africa</u>

Kirsten, J.F., Zyl, J., and Rooyen, C. (1994). South African Agriculture in the 1980s. Economic History of Developing Regions. *South African Journal of Economic History*, vol. 9, no. 2, pp. 19-48.

Kruger, A.C., and Nxumalo, M.P. (2017). Historical rainfall trends in South Africa: 1921-2015. *Water SA*, vol. 43, no. 2, pp. 285-297.

Lee, Y., and Liu, W. (2014). Forecasting Value of Agricultural Imports using a Novel two-stage Hybrid Model. *Computers and Electronics in Agriculture*, vol. 104, no. 2014, pp. 71-83.

Lencucha, R., Pal, N.E., and Appau, A. (2020). Government Policy and Agricultural Production: A Scoping Review to Inform Research and Policy on Healthy Agricultural Commodities. *Global Health*, vol. 16, no. 11, pp. 20-28.

Leonard, R. (1989). Resource Specialization, Population Growth, and Agricultural Production in the American Southwest. *American Antiquity*, vol. 54, no. 3, pp. 491-503.

Leppänen, S., Solanko, L., and Kosonen, R. (2017). The Impact of Climate Change on Regional Government Expenditures: Evidence from Russia. *Environmental Resource Economics*, vol. 7, no. 67, pp. 67–92.

Liebenberg, F., Pardey, P., and Kahn, M. (2010). South African Agricultural Research and Development: A Century of Change. *The University of Minnesota, Department of Applied Economics*, vol. 10, no. 1, pp. 20-25.

Liebenberg, F., Pardey, P.G., and Kahn, M. (2011). South African Agricultural R&D Investments: Sources, Structure, and Trends, 1910–2007. *Agrekon: Agricultural Economics Research, Policy and Practice in Southern Africa*, vol. 50, no. 2, pp.1-26. Lindersond, J. (2020). The Removal of Trade Sanctions' Impact on South African Wine Export. *Department of Economics*, vol. 1, no. 14, pp. 40-84.

Liu, W., Long, S., Wang, S., Tang, O., Hou, J., and Zhang, J. (2021). Effects of Smart Agricultural Production Investment Announcements on Shareholder Value: Evidence from China. *Journal of Management Science and Engineering,* vol. 21, no. 2, pp. 10-16.

Mabin, A.S., Nel, A., Thompson, L.M., Hall, M., Gordon, D.F., Lowe, C.C., Bundy, C.J., Cobbing, J.R.D., and Vigne, R. (2022). South Africa. *Encyclopedia Britannica*, viewed 23 Jan. 2022, <u>https://www.britannica.com/place/South-Africa</u>

MacDonald, GK., Brauman, KA., Sun, S., Carlson, K., Cassidy, ES., and Gerber, JS. (2015). Rethinking Agricultural Trade Relationships in an Era of Globalization. *BioScience*, vol. 65, no. 3, pp. 275–89.

MackKellar, N., New, M.G., and Christopher, J. (2014). Observed and Modelled Trends in Rainfall and Temperature for South Africa: 1960-2010. *South African Journal of Science*, vol. 110, no.7, pp.1-13.

Mahdi, E. (2016). Portmanteau Test Statistics for Seasonal Serial Correlation in Time Series Models. *Springer Plus* 5, viewed 20 January 2022, <u>https://doi.org/10.1186/s40064-016-3167-4</u>

Maïga, A., Bamba, A., Sy, B., Keita, G.H., Mouleye, I., and Diallo, M. Analysis of the Effects of Public Expenditure on Agricultural Growth in Mali. *Asian Journal of Agricultural Extension, Economics and Sociology*, vol. 39, no. 7, pp. 42-50.

Mainali, R. (2021). Spatial Fiscal Interactions in Colombian Municipalities: Evidence from Oil Price Shocks. *Journal of Risk and Financial Management*, vol. 14, no. 6, pp. 248-263.

Makin, A.J. (2015). Expansionary Versus Contractionary Government Spending. *Contemporary Economic Policy*, vol. 33, no. 1, pp. 56-65.

Maluleke, G. (2018). The Empirical Analysis of the Determinants of Government Expenditure in South Africa from 1970 to 2016. *Euro Economica*, vol. 37, no. 3, pp. 191-201.

Manuel, J. (2018). Wagner's law: A Revision and a New Empirical Estimation. *Hacienda Publica Espanola*, vol. 1, no. 10, pp, 77-82.

Mapfumo, A., Mushunje, A., and Chidoko, C. (2012). The Impact of Government Agricultural Expenditure on Poverty in Zimbabwe. *Russian Journal of Agricultural and Socio-Economic Sciences*, vol. 7, no. 7, pp. 16-22.

Matchaya, G.C. (2020). Public Spending on Agriculture in Southern Africa: Sectoral and Intra-Sectoral Impact and Policy Implications. *Journal of Policy Modelling*, vol. 42. No. 6. Pp. 1228-1247.

Mdluli, P., Mncayi, P., and Camel, T. (2019). Examining Factors that Drive Government Spending in South Africa. International Institute of Social and Economic Science, viewed 19 February 2022, <u>https://www.iises.net/proceedings/international-academic-conference-barcelona/front-page</u>

Meijerink, G., and Roza, P. (2007). The role of Agriculture in Economic Development. Markets, Chains and Sustainable Development Strategy and Policy Paper 4, viewed 5 April 2022, <u>https://edepot.wur.nl/690</u>

Meniago, C., and Petersen, M. (2013). What Causes Household Debt to Increase in South Africa? *Economic Modelling*, vol. 33. No. 4, pp. 482-492.

Milanzi, Mursali. (2012). Export Barrier Perceptions in Tanzania: The Influence of Social Networks. *Journal of African Business*, vol. 13, no. 1, pp. 29-39.

Mile, B.N., Ijirshar, V.U., Asom, S.T., Sokpo, J.T., and Fefa, J. (2021). Empirical Analysis of Government Agricultural Spending and Agricultural Output in Nigeria. *Asian Journal of Economics, Business, and Accounting*, vol. 21, no. 7, pp. 7-23.

Mogues, T., and Anson, J. (2018). How Comparable are Cross-Country Data on Agricultural Public Expenditures? *Global Food Security*, vol. 16, no. 2018, pp. 46-53.

Mohammed, G., Samir, G., Mostéfa, B. (2017). Agriculture Gross Production Value-Arable Land-Active Population in Agricultural Sector and Energy Consumption Causality in 76 countries: A Dynamic Panel Data Approach. *New Trends and Issues Proceedings on Humanities and Social Sciences*, vol. 4, no. 10, pp. 406-416. Molefe, K., and Choga, I. (2017). Government Expenditure and Economic Growth in South Africa: A Vector Error Correction Modelling and Granger Causality Test. *Journal of Economics and Behavioural Studies, AMH International*, vol. 9, no. 4, pp. 164-172.

Moreno-Dodson, B. (2008). Assessing the Impact of Public Spending on Growth and Empirical Analysis for Seven Fast-Growing Countries. Policy Research Working Paper. The World Bank. Paper no. 4663, viewed 26 April 2021, <u>http://hdl.handle.net/10986/6850</u>

Moritz, L. (1994). Trade and Industrial Policies in the New South Africa. Research Report no. 97, viewed 19 January 2021, <u>https://www.files.ethz.ch/isn/97899/97.pdf</u>

Morris, F., Warburton Toucher, M.L., Clulow, A., Kusangaya, S., Morris, C., and Bulcock, H. (2016). Improving the Understanding of Rainfall Distribution and Characterisation in the Cathedral Peak Catchments using a Geo-Statistical Technique. *Creative Commons Attribution*, vol. 42, no. 4, pp. 684-693.

Muhammad-Lawal, A., and Atte, A. (2009). An Analysis of Agricultural Production in Nigeria. *African Journal of General Agriculture*, vol. 2, no. 1, pp. 77-82.

Mutaudeen, O., and Hussainatu, A. (2014). Macroeconomic Policy and Agricultural Output in Nigeria: Implications for food security. *American journal of economics*, vol. 4, no. 20, pp. 99-113.

National Treasury. (2021). Budget review 2021, viewed 5 April 2022, http://www.treasury.gov.za/documents/national%20budget/2021/review/FullBR.pdf

Nguyen, T.A.A., and Luong, T.T.H. (2021). Fiscal Policy, Institutional Quality, and Public Debt: Evidence from Transition Countries. *Sustainability*, vol. 13, no. 19, pp. 10706-10720.

Nworji, I., Okwu, A., Obiwuru, T., and Nworji, L. (2012). Effect of Public Expenditure on Economic Growth in Nigeria: A Disaggregated Time Series Analysis. *International Journal of Management Sciences and Business Research*, vol. 1, no. 7, pp. 1-15.

Nyasha, S., and Odhiambo, N.M. (2019). Government Size and Economic Growth: A Review of International Literature. *SAGE Open*, vol. 9, no. 3, pp. 2-8.

Oberč, B.A., and Schnell, A.A. (2020). Approaches to Sustainable Agriculture: Exploring the Pathways towards the Future of Farming, *International Union for*

Conservation of Nature, viewed 6 April 2022, https://portals.iucn.org/library/efiles/documents/2020-017-En.pdf

Odhiambo, N. (2015). Government Expenditure and Economic Growth in South Africa: An Empirical Investigation. *Atlantic Economic Journal, Springer; International Atlantic Economic Society*, vol. 43, no. 3, pp. 393-406.

Ogiogio, G.O. (1995). Government Expenditure and Economic Growth in Nigeria. *Journal of Economic Management*, vol. 2, no. 1, pp. 05-10.

Oinam, K.S., Laishrum, P., and Kamal, V. (2022). The Impact of Public Expenditure on Agricultural Output: Empirical Evidence from Punjab, India. *Agricultural Economics Research Review*, vol. 2021, no. 34, pp. 157-164.

Okpara, C.S. (2017). Government Expenditure on Agriculture and Agricultural Output on Nigeria Economic Growth (1980-2015). *Middle East Journal of Scientific Research*, vol. 25, no. 5, pp. 1063-1079.

Oladele, M.F., Mah, G., and Mongale, I. (2017). The Role of Government Spending on Economic Growth in a Developing country. *Risk Governance and Control: Financial Markets and Institutions*, vol. 7, no. 2, pp. 140-146.

Olatunji, GB., Omotesho, O., Ayinde, O., and Adewumi, M. (2012). Empirical Analysis of Agricultural Production and Inflation Rate in Nigeria (1970-2006). *Agrosearch*, vol. 12, no. 2, pp. 101-106.

Olulu, R., Erhieyovwe, E., and Andrew, U. (2014). Government Expenditures and Economic Growth: The Nigerian Experience. *Mediterranean Journal of Social Sciences*, vol. 5, no. 10, pp. 89-94.

Oluwatoba, F.A., Ogundeji, A., and Willemse, B.J. (2015). Establishing the Linkages Between the South African Agricultural Trade Balance and Macroeconomic Indicators. *Agrekon*, vol. 53, no. 4, pp. 92-105.

Omabitan, O., and Khanal, A.R. (2022). Examining Farm Financial Management: How Do Small US Farms Meet Their Agricultural Expenses? *Journal of Risk and Financial Management*, vol. 15, no. 3, pp. 133-149.

Oosthuizen, M. (2007). Consumer Price Inflation Across the Income Distribution in South Africa. Development Policy Research Unit, working paper 27/129, viewed 19 January 2022, <u>http://www.dpru.uct.ac.za/sites/default/files/DPRU%20WP07-129.pdf</u>

Organisation for Economic Co-operation and Development (OECD). (2020). Producer and Consumer Support Estimates. OECD Agriculture statistics (database), viewed 03 May 2021, <u>http://dx.doi.org/10.1787/agr-pcse-data-en</u>

Pawlak, K., and Kołodziejczak, M. (2020). The Role of Agriculture in Ensuring Food Security in Developing Countries: Considerations in the Context of the Problem of Sustainable Food Production. *Sustainability*, vol. 2, no. 12, pp. 1-20.

Peña, D., and Rodríguez, J. (2006). The Log of the Determinant of the Autocorrelation Matrix for Testing Goodness of Fit in Time Series. *Journal of Statistical Planning and Inference*, vol. 136, no. 8, pp. 2706-2718.

Potelwa, X., Lubinga, M., and Ntshangase, T. (2016). Factors Influencing the Growth of South Africa's Agricultural Exports to World Markets. *European Scientific Journal*, vol. 16, no. 24, pp.195-204.

Praburaj, L. (2018). Role of Agriculture in the Economic Development of a Country. *Shanlax International Journal of Commerce*, vol. 6, no. 3, pp. 1-5.

Quiggin, J. (2008). Managing the Murray-Darling Basin: Some Implications for Climate Change Policy. *Economic Papers*, vol. 27, no. 2, pp. 160-166.

Rahman, S., and Anik, AF. (2020). Productivity and Efficiency Impact of Climate Change and Agroecology on Bangladesh Agriculture. *Land Use Policy*, vol. 94, no. 10, pp. 101-108.

Rasheed, H., and Tahir, M. (2012). FDI and Terrorism: Co-integration and Granger Causality. *International Affairs and Global Strategy*, vol. 4, no. 0, pp.1-5.

Rena, R., and Msoni, M. (2014). Global Financial Crises and its Impact on the South African Economy: A Further Update. *Journal of Economics*, vol. 5, no. 1, pp. 17-25.

Ribba, A. (2022). Monetary Policy Shocks in Open Economies and the Inflation Unemployment Trade-Off: The Case of the Euro Area. *Journal of Risk and Financial Management*, vol. 15, no. 4, pp. 146-165.

Rizwan, A., Jolita, V., Dalia, S., and Majid, F. (2017). Multivariate Granger Causality Between Macro Variables and KSE 100 Index: Evidence from Johansen cointegration and Toda and Yamamoto Causality. *Economic Research*, vol. 30, no. 13, pp. 1-29.

Ronayne, D. (2011). Which Impulse Response Function? Working Paper, no. 971. *Warwick Economic Research Papers*, viewed 20 February 2022, <u>https://www.researchgate.net/publication/241769290_Which_Impulse_Response_Fu</u> <u>nction</u>

Safi, S.K., and Al-Reqep, A.A. (2014). Comparative Study of Portmanteau Tests for the Residual's Autocorrelation in ARMA Models. *Science Journal of Applied Mathematics and Statistics*, vol. 2, no. 1, pp. 1-13.

Sangkhaphan, S., and Shu, Y. (2020). The Effect of Rainfall on Economic Growth in Thailand: A Blessing for Poor Provinces. *Economies*, vol. 8, no. 1, pp. 1-17.

Schneider, U.A., Havlík, P., Schmid, E., Valin, H., Mosnier, H., Obersteiner, M., Böttcher, H., Skalský, R., Balkovič, J., Sauer, T., and Fritz, S. (2021). Impacts of Population Growth, Economic Development, and Technical Change on Global Food Production and Consumption. *Agricultural Systems*, vol. 104, no. 2, pp. 204-215.

Schulze, R.E. (2005). Setting the Scene: The Current Hydroclimatic Landscape in Southern Africa. Water Research Commission, Pretoria, South Africa.

Selvanathan, E.A., Selvanathan, S., and Jayasinghe, M.S. (2021). Revisiting Wagner's and Keynesian's Propositions and the Relationship between Sectoral Government Expenditure and Economic Growth. *Economic Analysis and Policy,* vol. 71, no. 2021, pp. 355-370.

Selvaraj, KN. (1993). Impact of Government Expenditure on Agriculture and Performance of Agricultural Sector in India. *Bangladesh Journal of agricultural economics,* vol. 16, no. 2, pp. 13-20.

Setshedi, C., and Mosikari, T.J. 2019. Empirical Analysis of Macroeconomic Variables towards Agricultural Productivity in South Africa. *Italian Review of Agricultural Economics*, vol. 74, no. 2, pp. 3-15.

94

Shafuda, C. (2015). An Examination of the Relationship Between Government Spending and Economic Growth in Namibia, viewed 17 April 2022, https://repository.unam.edu.na/handle/11070/1471

Sharma, S., and Singh, S. (2020). Evidence of Wagner's Law in India Using Time Series from 1960 to 2018. *Current Strategies in Economics and Management*, vol. 1, no. 10, pp. 22-35.

Sholongo, F. (2019). An Econometric Analysis of the Impact of Imports on Inflation in Namibia, viewed 23 March 2022, https://uir.unisa.ac.za/bitstream/handle/10500/26869/dissertation_shilongo_f.pdf?seq uence=1&isAllowed=y

Single, N., and Beag, F.A. (2014). Cointegration Causality and Impulse Response Analysis in Major Apple Markets in India. *Agricultural economic review*, vo. 27, no. 2, pp. 289-298.

South African Reserve Bank (SARB). (2021). Online statistics: Republic of South Africa, viewed 04 August 2021, <u>https://www.resbank.co.za/en/home/what-we-do/statistics/releases/online-statistical-query</u>

South African Weather Service (SAWS). (2016). General Information South Africa -Annual Total Rainfall. Public Document, CLS-CI-INFO-110.1

Statistics South Africa (STATS SA). (2020). Census of Commercial Agriculture 2017 Report, viewed 4 May 2021, http://www.statssa.gov.za/?p=13144

Syed, W.G. (2015). The Impact of Agricultural Imports and Exports on Agricultural Productivity. *Journal of Economics and Sustainable Development*, vol. 6, no. 11, pp. 2222-2231.

Tabosa, F.J.S. (2021). Climate Sensitivity and the Value of Agricultural Production in the Brazilian Northeast: An Approach Using Spatial Panel Data. *International Journal of Business Administration*, vol. 12, no. 5, pp. 65-84.

Taiba, A. (2014). The Effectiveness of the Bank of Algeria to Sterilize the Accumulation of the Foreign-Exchange Reserves During the Period 2000-2011. *Academic Review of Social and Human Studies*, vol. 6, no. 2, pp. 26-34.
Tennant, W., and Hewitson, B. (2002). Intra-Seasonal Rainfall Characteristics and their Importance to the Seasonal Prediction Problem. *International Journal of Climatology*, vol. 22, no. 9, pp. 1033-1048.

Tiwari, Y.K., Revadekar, J.W., and Kumar, K.R. (2013). Variations in Atmospheric Carbon Dioxide and its Association with Rainfall and Vegetation over India. *Atmospheric Environment*, vol. 68, no. 40, pp. 45-51.

Torres, M., Howitt, R., and Rodrigues, L. (2019). Analysing Rainfall Effects on Agricultural Income: Why Timing Matters. *EconomiA*, vol. 20, no. 1, pp. 1-14.

Tsakok, I., and Bruce, G. 2007. Agriculture in Economic Development: Primary Engine of Growth or Chicken and Ggg? *American Journal of Agricultural Economics*, vol. 8, no. 5, pp. 1145-1151.

Ullah, R., and Shivakoti G. (2018). Redefining Diversity and Dynamics of Natural Resources Management in Asia. *Sustainable Natural Resources Management in Dynamic Asia*, vol. 01, no. 20, pp. 371-385.

United Nations (UN). (2019). World Population Prospects, viewed 29 July 2021, https://population.un.org/wpp/Publications/Files/WPP2019_Highlights.pdf

Uremadu1, S.O., Ariwa, F.O., and Duru-Uremadu, C.E. (2018). Impact of Government Agricultural Expenditure on Agricultural Productivity in Nigeria. *Current Investigations in Agriculture and Current Research*, vol. 5, no. 3, viewed 10 December 2021, <u>https://lupinepublishers.com/agriculture-journal/pdf/CIACR.MS.ID.000215.pdf#pae=1</u>

Vasić, N., Kilibarda., M., and Kaurin, T. (2019). The Influence of Online Shopping Determinants on Customer Satisfaction in the Serbian Market. *Journal of theoretical and applied electronic commerce research*, vol. 14, no. 2, pp. 70-89.

Vink, N., and Van Rooyen, J. 2009. The Economic Performance of Agriculture in South Africa since 1994: Implications for Food Security. Development Planning Division Working Paper Series no.17.

Visser, M., and Ferrer. S. (2015). Farm Workers' Living and Working Conditions in South Africa: Key Trends, Emergent Issues, and Underlying and Structural Problems. The Pretoria Office of the International Labour Organisation Report, viewed 7 April 2022,

https://www.ilo.org/wcmsp5/groups/public/---

africa/documents/publication/wcms_385959.pdf

Von Loeper, W.J., and Blignaut, J. (2018). The Struggles of Smallholder Farmers: A Cause of Modern Agricultural Value Chains, IntechOpen partners, South Africa.

Wagner, A. (1876). Three Extracts on Public Finance. In Classics in the Theory of Public Finance. Edited by Richard A. Musgrave and Alan T. Peacock. New York: St. Martin's Press.

Wielechowski, M. (2019). Government Expenditure on Agriculture. *European, Regional and World Perspective*, vol. 21, no. 10, pp. 561-570.

World bank (WB). (2015). The South Africa Agriculture Public Expenditure Review, viewed 08 August 2021, <u>https://openknowledge.worldbank.org/bitstream</u>

World bank (WB). (2020). International Labour Organisation, ILOSTAT database, viewed 02 May 2021, https://data.worldbank.org/indicator/SL.AGR.EMPL.ZS

Wu, S., Tang, J., and Lin, E.S. (2010). The Impact of Government Expenditure on Economic Growth: How Sensitive to the Level of Development? *Journal of Policy Modelling*, vol. 32, no. 6, pp. 804-817.

Zainab, NT., and Umar, S. (2015). The Impact of Macroeconomic Variables on Agricultural Productivity in Malaysia. *Southeast Asia Journal of contemporary business, economics, and law*, vol. 8, no. 3, pp. 2289-1560.

Zavale, H., Mlay, G., Boughton, D., Chamusso, A., and Chilonda, P. (2011). The Structure and Trends of Public Expenditure on Agriculture in Mozambique. Regional Strategic Analysis and Knowledge Support System, Working Paper No. 34, viewed 5 April 2022, <u>https://core.ac.uk/download/pdf/132651428.pdf</u>

Zeraibi, A., and Mivumbi, M. (2019). The Impact of Public Expenditure on the Agricultural Sector Productivity in China. *American Journal of Humanities and Social Sciences Research*, vol. 3, no. 9, pp. 173-180.

Zhu, G., Du, Z., and Escanciano, J.C. (2017). Automatic Portmanteau Tests with Applications to Market Risk Management. *The Stata Journal*, vol. 17, no. 4, pp. 901-915.

APPENDIXES

APPENDIX 1: DATA

Year	VAP	GEA	AAR	СРІ	FIV	PG
1983	7881,4	833	373,4	9,616959	1073,9	30993,76
1984	8774,4	1220	400,9	10,72546	1682	31841,59
1985	10933,9	1565	504,5	12,47309	1298,3	32678,87
1986	11381	1706	428,8	14,79993	1473,4	33495,95
1987	13348,8	1571	471,9	17,19169	1511,1	34297,73
1988	16175,9	2058	592 <i>,</i> 4	19,38871	2081,4	35100,91
1989	19334,8	2110	513,4	22,24483	2079	35930,05
1990	20198	2055	388,8	25,43051	1936,2	36800,51
1991	21952,4	2240	506,7	29,33023	2257,8	37718,95
1992	22126,8	2817	314,2	33,39971	4477,9	38672,61
1993	25723,7	5868	450	36,64531	3813,1	39633,75
1994	28162,4	3717	389,7	39,92086	4894,9	40564,06
1995	29941,3	3239	501,4	43,38616	6834,3	41435,76
1996	36519,3	3651	567,4	46,57684	7745,2	42241,01
1997	40458,3	3623	486,2	50,58141	8641,5	42987,46
1998	43117,4	3885	495,6	54,06169	9401,6	43682,26
1999	44669,7	3746	435,7	56,86289	8963,8	44338,54
2000	46660,9	3743	624,4	59,89877	9644,1	44967,71
2001	52185,6	4224	538,5	63,31414	10704,4	45571,27
2002	68283	5190	408,9	69,32563	15162,6	46150,91
2003	72931,5	5855	360,5	73,26293	13910,5	46719,2
2004	73583,5	7192	467,7	72,75593	16415,5	47291,61
2005	77112,6	6806	384,9	74,25677	16286,4	47880,6
2006	79152,1	9156	580,6	76,66559	20588,5	48489,46
2007	98535,1	10221	409,1	81,40185	29304,5	49119,76
2008	129164,3	13418	435,7	89,58703	38427,5	49779,47

2009 134566,9 14639 485,8 96,09514 35039,2	2 50477,01
2010 132654,9 13591 477,6 100 39531,6	5 51216,96
2011 144588,2 15868 557,9 105,0172 50798,7	7 52003,76
2012 168591,1 17082 378 111,0283 59563,5	5 52832,66
2013 184623,9 18708 399,4 117,4417 64251,9	9 53687,12
2014 209472,3 20219 461,6 124,648 67243,2	2 54544,19
2015 224792 20159 338,4 130,2686 76888,8	8 55386,37
2016 244001,2 20330 405,8 138,8593 92203,9	9 56207,65
2017 269413,1 21785 415,8 146,0537 87630	57009,76
2018 286095,9 20727 383 152,6328 87004,8	8 57792,52
2019 284383,2 22289 381,9 158,9279 91688,9	9 58558,27

APPENDIX 2: DESCRIPTIVE STATISTICS

	VAP	GEA	CPI	AAR	FIV	PG
Mean	91391.64	8570.432	69.29939	451.7973	26823.08	45083.78
Median	52185.60	5190.000	63.31414	435.7000	10704.40	45571.27
Maximum	286095.9	22289.00	158.9279	624.4000	92203.90	58558.27
Minimum	7881.400	833.0000	9.616959	314.2000	1073.900	30993.76
Std. Dev.	86601.48	7291.566	43.95397	75.11357	30449.67	8035.092
Skewness	1.017649	0.713730	0.446555	0.435300	1.065556	-0.086326
Kurtosis	2.759983	1.927244	2.131203	2.461838	2.669251	1.929739
Jarque-Bera	<mark>6.475070</mark>	4.915526	2.393365	1.614990	7.170348	1.811870
Probability	0.039261	0.085626	0.302195	0.445974	0.027732	0.404164
Sum	3381491.	317106.0	2564.078	16716.50	992453.9	1668100.
Sum Sq. Dev.	2.70E+11	1.91E+09	69550.24	203113.7	3.34E+10	2.32E+09
Observations	37	37	37	37	37	37

APPENDIX 3: JOHANSEN COINTEGRATION (ALL VARIABLES)

Date: 12/12/21 Time: 00:52 Sample (adjusted): 1986 2019 Included observations: 34 after adjustments Trend assumption: Linear deterministic trend Series: VAP GEA CPI AAR FIV PG Lags interval (in first differences): 1 to 2

Hypothesized No. of CE(s)	Eigenvalue	Trace Statistic	0.05 Critical Value	Prob.**
None *	0.938838	192.0768	95.75366	0.0000
At most 1 *	0.708963	97.07278	69.81889	0.0001
At most 2 *	0.551569	55.10638	47.85613	0.0090
At most 3	0.424873	27.83837	29.79707	0.0827
At most 4	0.226034	9.030797	15.49471	0.3625
At most 5	0.009340	0.319057	3.841465	0.5722

Unrestricted Cointegration Rank Test (Trace)

Trace test indicates 3 cointegrating eqn(s) at the 0.05 level

* denotes rejection of the hypothesis at the 0.05 level

**MacKinnon-Haug-Michelis (1999) p-values

Unrestricted Cointegration Rank Test (Maximum Eigenvalue)

Hypothesized No. of CE(s)	Eigenvalue	Max-Eigen Statistic	0.05 Critical Value	Prob.**
None *	0.938838	95.00406	40.07757	0.0000
At most 1 *	0.708963	41.96640	33.87687	0.0044
At most 2	0.551569	27.26801	27.58434	0.0548
At most 3	0.424873	18.80757	21.13162	0.1026
At most 4	0.226034	8.711740	14.26460	0.3109
At most 5	0.009340	0.319057	3.841465	0.5722

Max-eigenvalue test indicates 2 cointegrating eqn(s) at the 0.05 level

* denotes rejection of the hypothesis at the 0.05 level

**MacKinnon-Haug-Michelis (1999) p-values

APPENDIX 4: JOHANSEN COINTEGRATION FOR VAP AND GEA

Date: 02/19/22 Time: 18:22 Sample (adjusted): 1986 2019 Included observations: 34 after adjustments Trend assumption: Linear deterministic trend Series: VAP GEA Lags interval (in first differences): 1 to 2

Hypothesized No. of CE(s)	Eigenvalue	Trace Statistic	0.05 Critical Value	Prob.**
None *	0.465245	21.61477	15.49471	0.0053
At most 1	0.009735	0.332611	3.841465	0.5641

Unrestricted Cointegration Rank Test (Trace)

Trace test indicates 1 cointegrating eqn(s) at the 0.05 level

* denotes rejection of the hypothesis at the 0.05 level

**MacKinnon-Haug-Michelis (1999) p-values

Unrestricted Cointegration Rank Test (Maximum Eigenvalue)

Hypothesized No. of CE(s)	Eigenvalue	Max-Eigen Statistic	0.05 Critical Value	Prob.**
None *	0.465245	21.28216	14.26460	0.0033
At most 1	0.009735	0.332611	3.841465	0.5641

Max-eigenvalue test indicates 1 cointegrating eqn(s) at the 0.05 level

* denotes rejection of the hypothesis at the 0.05 level

**MacKinnon-Haug-Michelis (1999) p-values

APPENDIX 5: GRANGER CAUSALITY

Pairwise Granger Causality Tests Date: 12/12/21 Time: 00:59 Sample: 1983 2019 Lags: 2

Null Hypothesis:	Obs	F-Statistic	Prob.
DGEA does not Granger Cause DVAP	34	0.24541	0.7840
DVAP does not Granger Cause DGEA		1.59284	0.2206
DCPI does not Granger Cause DVAP	34	0.39003	0.6805
DVAP does not Granger Cause DCPI		2.08565	0.1425
DAAR does not Granger Cause DVAP	34	0.42173	0.6599
DVAP does not Granger Cause DAAR		2.13975	0.1359
DFIV does not Granger Cause DVAP	34	9.92527	0.0005
DVAP does not Granger Cause DFIV		1.46711	0.2472
DPG does not Granger Cause DVAP	34	0.54354	0.5865
DVAP does not Granger Cause DPG		0.29980	0.7432

APPENDIX 6: LOG TRANSFORMED DATA

YEAR	LNVAP	LNGEA	LNCPI	LNAAR	LNFIV	LNPG
1983	8.972261	6.725034	2.263528	5.922650	6.979052	10.34154
1984	9.079594	7.106606	2.372620	5.993712	7.427739	10.36853
1985	9.299623	7.355641	2.523573	6.223568	7.168811	10.39448
1986	9.339701	7.441907	2.694622	6.060991	7.295328	10.41918
1987	9.499182	7.359468	2.844426	6.156767	7.320593	10.44283
1988	9.691278	7.629490	2.964691	6.384182	7.640796	10.46598
1989	9.869662	7.654443	3.102110	6.241055	7.639642	10.48933
1990	9.913339	7.628031	3.235950	5.963065	7.568483	10.51327
1991	9.996632	7.714231	3.378619	6.227919	7.722146	10.53792
1992	10.00454	7.943428	3.508547	5.750030	8.406909	10.56289
1993	10.15517	8.677269	3.601285	6.109248	8.246198	10.58744
1994	10.24574	8.220672	3.686899	5.965377	8.495949	10.61064
1995	10.30699	8.083020	3.770141	6.217404	8.829709	10.63190
1996	10.50560	8.202756	3.841103	6.341065	8.954829	10.65115
1997	10.60803	8.195058	3.923584	6.186620	9.064331	10.66866
1998	10.67168	8.264878	3.990126	6.205769	9.148635	10.68470
1999	10.70705	8.228444	4.040643	6.076954	9.100950	10.69961
2000	10.75066	8.227643	4.092656	6.436791	9.174102	10.71370
2001	10.86256	8.348538	4.148109	6.288787	9.278410	10.72703
2002	11.13142	8.554489	4.238815	6.013471	9.626587	10.73967
2003	11.19728	8.675051	4.294055	5.887492	9.540399	10.75191
2004	11.20618	8.880725	4.287110	6.147827	9.705981	10.76409
2005	11.25302	8.825560	4.307529	5.952984	9.698086	10.77647
2006	11.27913	9.122165	4.339453	6.364062	9.932488	10.78910
2007	11.49817	9.232200	4.399398	6.013960	10.28550	10.80202
2008	11.76884	9.504352	4.495211	6.076954	10.55653	10.81536
2009	11.80982	9.591444	4.565339	6.185797	10.46422	10.82927
2010	11.79551	9.517163	4.605170	6.168774	10.58486	10.84383
2011	11.88164	9.672060	4.654124	6.324180	10.83563	10.85907
2012	12.03523	9.745781	4.709785	5.934894	10.99480	10.87488
2013	12.12608	9.836707	4.765942	5.989963	11.07057	10.89093
2014	12.25235	9.914378	4.825494	6.134699	11.11607	10.90677
2015	12.32293	9.911406	4.869599	5.824229	11.25012	10.92209
2016	12.40493	9.919853	4.933461	6.005860	11.43176	10.93681
2017	12.50400	9.988977	4.983975	6.030204	11.38088	10.95098
2018	12.56408	9.939192	5.028035	5.948035	11.37372	10.96461
2019	12.55808	10.01185	5.068451	5.945159	11.42616	10.97778

APPENDIX 7: VAR RESULTS

	LNVAP	LNGEA	LNCPI	LNAAR	LNFIV	LNPG
INVAP(-1)	0.638779	-0 267575	0.061897	-0.889785	-0 167321	0 002949
,	(0.21171)	(0.56009)	(0.04833)	(0.52025)	(0.54881)	(0.00156)
	[3.01730]	[-0.47773]	[1.28081]	[-1.71029]	[-0.30488]	[1.89579]
LNVAP(-2)	-0.073059	0.576168	-0.064910	0.332941	-0.282469	0.004210
	(0.16780)	(0.44393)	(0.03830)	(0.41235)	(0.43499)	(0.00123)
	[-0.43540]	[1.29700]	[-1.09402]	[0.80742]	[-0.04937]	[3.41433]
LNGEA(-1)	0.161955	0.748791	0.031194	-0.360752	0.194284	0.001810
,	(0.07776)	(0.20573)	(0.01775)	(0.19110)	(0.20159)	(0.00057)
	[2.08264]	[3.63960]	[1.75730]	[-1.88777]	[0.96376]	[3.16678]
LNGEA(-2)	-0.033866	-0.137518	-0.014016	0.498465	0.297081	-0.000738
	(0.06930)	(0.16333) [-0.75010]	(0.01562) [-0.88602]	(0.17029)	(0.17964)	(0.00051)
	[-0.40070]	[-0.75010]	[-0.00002]	[2:52710]	[1.00070]	[-1.4070]
LNCPI(-1)	-0.208038	1.150326	0.988190	0.478136	-0.119004	-0.003326
	(0.61726)	(1.63303)	(0.14090)	(1.51686)	(1.60013)	(0.00454)
	[-0.33704]	[0.70441]	[7.01332]	[0.31521]	[-0.07437]	[-0.73337]
	0 564057	1 590250	0 272909	0 165919	0 950179	0.004707
LINCFI(-2)	(0.62524)	(1 65414)	-0.273898	(1 53647)	(1 62082)	(0.004707
	[-0.90359]	[-0.96084]	[-1.91908]	[0.10792]	[-0.53009]	[1.02457]
		· · · · ·				
LNAAR(-1)	0.088903	-0.235957	0.033549	-0.119990	0.229443	0.000604
	(0.06601)	(0.17463)	(0.01507)	(0.16221)	(0.17111)	(0.00049)
	[1.34689]	[-1.35120]	[2.22660]	[-0.73973]	[1.34090]	[1.24602]
INAAR(-2)	0 120143	0.083568	0 059577	-0 108306	0.007926	0.001190
	(0.07328)	(0.19387)	(0.01673)	(0.18008)	(0.18996)	(0.00054)
	[1.63955]	[0.43106]	[3.56168]	[-0.60145]	[0.04172]	[2.20971]
LNFIV(-1)	0.146890	0.481962	-0.023625	0.252164	0.354477	-0.000227
	(0.07002)	(0.18524)	(0.01598)	(0.17206)	(0.18151)	(0.00051)
	[2.09790]	[2.00102]	[-1.47810]	[1.40555]	[1.95294]	[-0.44191]
LNFIV(-2)	-0.234572	-0.517797	-0.043699	0.318777	-0.199583	-0.002100
	(0.07650)	(0.20240)	(0.01746)	(0.18800)	(0.19832)	(0.00056)
	[-3.06617]	[-2.55829]	[-2.50228]	[1.69561]	[-1.00636]	[-3.73591]
	4 207022	6 202455	E 0004E0	25 66700	11 50500	1 000005
LNFG(-1)	4.397032	-0.393433	(1 45569)	-35.007.09	(16 5314)	(0.04686)
	[0.68951]	[-0.37896]	[4.01057]	[-2.27598]	[0.70143]	[38.9035]
LNPG(-2)	1.527991	8.699318	-4.175405	29.76986	-0.373160	-0.860776
	(5.95163)	(15.7458)	(1.35859)	(14.6257)	(15.4286)	(0.04373)
	[0.25673]	[0.55249]	[-3.07334]	[2.03545]	[-0.02419]	[-19.6830]
С	-57.09542	-21.56729	-16.73817	68.24213	-109.0442	0.325364
	(17.9046)	(47.3688)	(4.08711)	(43.9993)	(46.4147)	(0.13156)
	[-3.18887]	[-0.45531]	[-4.09536]	[1.55098]	[-2.34935]	[2.47309]
D. a success of	0.007000	0.000475	0.000707	0.540004	0.000000	0.000000
Adi R-squared	0.997962	0.962175	0.999787	0.546631	0.992602	0.999996
Sum sq. resids	0.069386	0.485658	0.003616	0.419022	0.466289	3.75E-06
S.E. equation	0.056160	0.148578	0.012820	0.138009	0.145585	0.000413
F-statistic	906.8056	101.0160	8613.787	2.228383	252.8599	473379.8
Log likelihood	59.24689	25.19514	110.9496	27.77782	25.90736	231.2138
Akaike Alu Schwarz SC	-2.04268U	-0.696865	-5.59/120	-U.844447 -0.266746	-0.131563	-12.46936
Mean dependent	11.02903	8.743379	4.077658	6.108118	9,495147	10.71990
S.D. dependent	1.005709	0.895170	0.706922	0.165239	1.380309	0.168673
Determinant resid covariance	e (dof adj.)	1.80E-19				
Log likelihood	3	1.11E-20 506.0803				
Akaike information criterion		-24.46173				
Schwarz criterion		-20.99553				
Number of coefficients		78				

APPENDIX 8: STABILITY TEST RESULTS

Roots of Characteristic Polynomial Endogenous variables: LNVAP LNGEA LNCPI LNAAR LNFIV LNPG Exogenous variables: C Lag specification: 1 2 Date: 02/21/22 Time: 16:39

Root	Modulus
0.999311	0.999311
0.788846 - 0.194580i	0.812490
0.788846 + 0.194580i	0.812490
0.652225 - 0.407129i	0.768864
0.652225 + 0.407129i	0.768864
0.213037 - 0.719717i	0.750585
0.213037 + 0.719717i	0.750585
0.724670	0.724670
-0.522612	0.522612
-0.176970 - 0.446576i	0.480363
-0.176970 + 0.446576i	0.480363
0.277539	0.277539

No root lies outside the unit circle.

VAR satisfies the stability condition.



APPENDIX 9: AUTOCORRELATION RESULTS

System Residual Portmanteau Tests for Autocorrelations Null Hypothesis: no residual autocorrelations up to lag h Date: 12/12/21 Time: 01:12 Sample: 1985 2019 Included observations: 35

Lags	Q-Stat	Prob.	Adj Q-Stat	Prob.	df
1	32,41480	0.6399	33,36818	0.5944	36
2	67.29024	0.6351	70.35728	0.5328	72
3	99.21200	0.7154	105.2717	0.5564	108
4	141.1624	0.5513	152.6351	0.2953	144
5	165.2624	0.7775	180.7518	0.4702	180
6	201.9443	0.7451	225.0230	0.3227	216
7	219.0128	0.9343	246.3586	0.5884	252
8	242.5154	0.9760	276.8250	0.6714	288
9	268.8734	0.9886	312.3068	0.6696	324
10	298.2606	0.9923	353.4489	0.5874	360
11	317.0200	0.9986	380.8063	0.6996	396
12	348.1969	0.9988	428.2495	0.5419	432

*The test is valid only for lags larger than the System lag order. df is degrees of freedom for (approximate) chi-square distribution *df and Prob. may not be valid for models with lagged endogenous va...

APPENDIX 10: HETEROSCEDASTICITY TEST RESULTS

VAR Residual Heteroskedasticity Tests (Levels and Squares) Date: 02/21/22 Time: 16:27 Sample: 1983 2019 Included observations: 35

Joint test:		
Chi-sq	df	Prob.
525.1318	504	0.2491

Individual components:

Dependent	R-squared	F(24,10)	Prob.	Chi-sq(24)	Prob.
res1*res1	0.550876	0.511066	0.9136	19.28067	0.7369
res2*res2	0.896603	3.613121	0.0194	31.38112	0.1431
res3*res3	0.632086	0.715843	0.7596	22.12299	0.5719
res4*res4	0.853664	2.430666	0.0724	29.87825	0.1888
res5*res5	0.758568	1.309149	0.3391	26.54989	0.3259
res6*res6	0.810814	1.785753	0.1704	28.37849	0.2444
res2*res1	0.798966	1.655952	0.2049	27.96381	0.2616
res3*res1	0.624747	0.693694	0.7780	21.86613	0.5873
res3*res2	0.739617	1.183538	0.4074	25.88658	0.3589
res4*res1	0.725130	1.099203	0.4602	25.37956	0.3854
res4*res2	0.588160	0.595052	0.8563	20.58558	0.6630
res4*res3	0.841590	2.213636	0.0955	29.45565	0.2035
res5*res1	0.843626	2.247884	0.0914	29.52691	0.2009
res5*res2	0.903417	3.897421	0.0148	31.61960	0.1367
res5*res3	0.803871	1.707784	0.1903	28.13548	0.2544
res5*res4	0.774908	1.434428	0.2823	27.12178	0.2988
res6*res1	0.437260	0.323758	0.9886	15.30409	0.9115
res6*res2	0.839508	2.179519	0.0999	29.38278	0.2061
res6*res3	0.571103	0.554817	0.8851	19.98860	0.6974
res6*res4	0.762472	1.337509	0.3253	26.68651	0.3193
res6*res5	0.863823	2.643085	0.0559	30.23382	0.1771

APPENDIX 11: NORMALITY RESULTS

VAR Residual Normality Tests Orthogonalization: Cholesky (Lutkepohl) Null Hypothesis: Residuals are multivariate normal Date: 12/12/21 Time: 01:15 Sample: 1983 2019 Included observations: 35

Component	Skewness	Chi-sq	df	Prob.*
1	0.681677	2.710656	1	0.0997
2	0.439981	1.129235	1	0.2879
3	-0.026239	0.004016	1	0.9495
4	0.354811	0.734363	1	0.3915
5	0.300509	0.526782	1	0.4680
6	0.235003	0.322153	1	0.5703
Joint		5.427206	6	0.4903
Component	Kurtosis	Chi-sq	df	Prob.
1	2 529783	0 322443	1	0 5701
2	3.268545	0.105170	1	0.7457
3	2.466333	0.415335	1	0.5193
4	2.530958	0.320835	1	0.5711
5	4.046593	1.597395	1	0.2063
6	2.489641	0.379847	1	0.5377
Joint		3.141024	6	0.7910
Component	Jarque-Bera	df	Prob.	
1	3.033099	2	0.2195	
2	1.234405	2	0.5395	
3	0.419351	2	0.8108	
4	1.055198	2	0.5900	
5	2.124177	2	0.3457	
6	0.702001	2	0.7040	
Joint	8.568230	12	0.7393	

*Approximate p-values do not account for coefficient estimation